TUM School of Social Sciences and Technology TUM School of Management



Munich Papers in Political Economy Working Paper No. 04/2024

Trapped in Poverty: The Impact of Large-Scale Land Acquisitions on Education. A Geospatial Approach

Ben-Amon Kosbab

August 2024

Trapped in Poverty: The Impact of Large-Scale Land Acquisitions on Education. A Geospatial Approach

Ben-Amon Kosbabⁱ

Abstract

Large parts of the population in developing countries depend on agriculture for their income and food security. However, agriculture-dependent households are vulnerable to agricultural shocks, which prevent them from investing in education, thus hindering their socio-economic progress and their ability to reduce dependence on agriculture. Research on the impact of agricultural shocks on education predominantly focuses on those caused by extreme weather events and fluctuations in agricultural commodity prices. The impact of large-scale land acquisitions on education has not been studied, despite their growing number and potential to disrupt the agricultural production of small-scale farmers. This paper fills the research gap by hypothesizing that large-scale land acquisitions negatively impact the education of people in their vicinity due to resulting food insecurity and income loss, leading households to divert educational resources to basic needs and withdraw children from school to contribute to income. The negative impact on education is expected to be more pronounced for boys, who find rural employment more easily and are thus more frequently withdrawn from school. Employing a geospatial approach, this paper links 322 large-scale land acquisitions in Africa to 46,711 Afrobarometer respondents. The results of the regression analysis indicate that being affected by a large-scale land acquisition between the ages of 0 and 16 has a statistically significant negative impact on education. The hypothesized stronger negative impact on male education is not supported by the results. The findings imply that large-scale land acquisitions hinder rural development and entrench poverty, contrary to claims by investors and politicians.

Keywords: Education, large-scale land acquisition, agricultural shocks, rural development **JEL Codes:** 120, 124,125

ⁱ University of Konstanz & Universitat Pompeu Fabra Barcelona E-Mail: ben-amon.kosbab@uni-konstanz.de

Table of Contents

1.	Introduction1
2.	Literature Review
2.1.	Agricultural Shocks and Education4
2.2.	Determinants and Consequences of LSLAs5
3.	Theory7
3.1.	The Impact of LSLAs on Education7
3.2.	The Gendered Impact of LSLAs on Education8
4.	Research Design9
4.1.	Identification Strategy and Data Selection11
4.2.	Operationalization15
4.3.	Model Specification19
5.	Results
5. 6.	Results20Robustness Checks and Causal Mechanisms Testing23
5. 6. 7.	Results20Robustness Checks and Causal Mechanisms Testing23Discussion24
5. 6. 7. 8.	Results
5. 6. 7. 8. 9.	Results
5. 6. 7. 8. 9. Appe	Results20Robustness Checks and Causal Mechanisms Testing23Discussion24Conclusion26References28endix33
 5. 6. 7. 8. 9. Appe Appe 	Results20Robustness Checks and Causal Mechanisms Testing23Discussion24Conclusion26References28endix33endix A: Countries Included in the Analysis33
 5. 6. 7. 8. 9. Appe Appe Appe 	Results20Robustness Checks and Causal Mechanisms Testing23Discussion24Conclusion26References28endix33endix A: Countries Included in the Analysis33endix B: Regression Results33
 5. 6. 7. 8. 9. Appe Appe Appe 	Results20Robustness Checks and Causal Mechanisms Testing23Discussion24Conclusion26References28endix33endix A: Countries Included in the Analysis33endix B: Regression Results33endix C: Robustness Checks37
 5. 6. 7. 8. 9. Appe Appe Appe Appe 	Results20Robustness Checks and Causal Mechanisms Testing23Discussion24Conclusion26References28endix33endix A: Countries Included in the Analysis33endix B: Regression Results33endix C: Robustness Checks37endix D: Testing the Causal Mechanisms53

List of Tables

Table B.1: Regression Results	34
Table B.2: Regression Results (Interaction)	35
Table B.3: Regression Results (Gender Differentiated)	36
Table C.1: Comparative Regression Results	38
Table C.2: Comparative Regression Results (Interaction)	
Table C.3: Comparative Regression Results (Gender Differentiated)	40
Table C.4: Logit Regression Results Primary Education	41
Table C.5: Logit Regression Results Primary Education (Interaction)	42
Table C.6: Logit Regression Results Primary Education (Gender Differentiated)	43
Table C.7: Logit Regression Results Secondary Education	44
Table C.8: Logit Regression Results Secondary Education (Interaction)	45
Table C.9: Logit Regression Results Secondary Education (Gender Differentiated)	46
Table C.10: Regression Results Ethnicity	47
Table C.11: Regression Results Ethnicity (Interaction)	48
Table C.12: Regression Results Ethnicity (Gender Differentiated)	49
Table C.13: Regression Results Urban	50
Table C.14: Regression Results Urban (Interaction)	51
Table C.15: Regression Results Urban (Gender Differentiated)	52
Table D.1: Operationalization of Variables	53
Table D.2: Logit Regression Results School Infrastructure	56
Table D.3: Logit Regression Results No Income	57
Table D.4: Logit Regression Results Food Insecurity	58
Table D.5: Regression Results Causal Mechanism	59

List of Figures

Figure 1: Hypotheses	10
Figure 2: Identification Strategy	12
Figure 3: Locations of Afrobarometer Respondents	14
Figure 4: Locations of LSLAs	15
Figure 5: Histogramm of Education Levels	16
Figure 6: Regression Results	21
Figure 7: Marginal Effects of Interaction Term	22
Figure 8: Regression Analysis (Gender Differentiated)	23

1. Introduction

Education is a key determinant of socio-economic outcomes like income (Hofmarcher, 2021), political participation (Bömmel & Heineck, 2023), and health (Davies et al., 2023). Improving access to education is a central goal of the United Nations' 2030 Agenda for Sustainable Development (Filho et al., 2023). However, access to education in developing countries¹ remains limited, with approximately 260 million children eligible for primary and secondary schooling not enrolled. In sub-Saharan Africa, only two-thirds of children complete primary education. Additionally, developing countries are characterized by significant gender disparities in education that disproportionately disadvantage girls (BMZ, 2024).

For households whose livelihoods depend on agriculture, investing regularly and sustainably in education is particularly difficult. Agricultural shocks—disruptions in agricultural income and production (Alam et al., 2020)—lead to volatility in yields and earnings. Agriculture is the primary livelihood for large parts of the population in developing countries. In 2019, 48.1% of the working population in Africa and 29% in Asia were employed in the agricultural sector. More than 2 billion people in developing countries live in households whose livelihoods depend on agriculture (Davis et al., 2023). Given the importance of agriculture in developing countries, agricultural shocks prevent large segments of the population from improving their socio-economic situation by investing in education. This reinforces dependence on agriculture, trapping households in a cycle of poverty and potentially exacerbating existing educational gender disparities (Björkman-Nyqvist, 2013).

Research on the impact of agricultural shocks on education primarily focuses on agricultural shocks caused by weather and climate conditions, as well as those resulting from decreasing prices for agricultural commodities (Alam et al., 2020). So far, the literature has not addressed agricultural shocks caused by large-scale land acquisitions (LSLAs), defined as land purchases

¹ I use the term developing countries, because it is used in the literature on agricultural shocks. The term typically refers to low and middle-income countries, but there is no uniform definition, making it vague (Farias, 2023).

by investors² (D'Odorico et al., 2017). Investors increasingly buy land in developing countries that was previously used by small-scale farmers and secured the livelihoods of the rural population (Dell'Angelo et al., 2021). The small-scale farmers lose access to the land, leading to disruptions in agricultural income and production. This has far-reaching consequences for the affected communities and farmers, causing income loss (Yengoh & Armah, 2015), food insecurity (Castet, 2024), and conflicts (Balestri & Maggioni, 2021). The number of LSLAs is increasing rapidly. Investors have globally purchased an estimated 90 million hectares of fertile land since the early 2000s. This land has the capacity to provide food for up to 330 million people (Müller et al., 2021). Most of the purchased land is used for export-oriented agriculture or land speculation (Dell'Angelo et al., 2017).

Given the significance of education as a key socioeconomic predictor, the extensive dependence on agricultural livelihoods in developing countries, and the increasing number of LSLAs, this paper addresses the following research question: *What is the impact of LSLAs on education*?

As my primary hypothesis, I argue that LSLAs negatively affect the education of people living in their vicinity through two mechanisms: loss of income and food insecurity. Once the purchased land is put into production, LSLAs deprive individuals of land access, preventing them from growing crops. This results in increased food insecurity and decreased income. Consequently, households redirect funds from education to cover essential needs, thereby reducing their educational investments. Additionally, it compels households to withdraw their children from school so they can contribute to household income through work. As a secondary hypothesis, I argue that the detrimental effects of LSLAs on education are more pronounced for males than for females. Since it is easier for males to find employment in rural areas, they are more likely to be withdrawn from school.

To test the hypotheses, I employ a geospatial research design. My analysis is based on 322 LSLAs in Africa from the Land Matrix database and 46,711 respondents from the sixth round

² The term large-scale land acquisitions is the most commonly used term in literature dealing with land deals. Some papers refer to it as land grabbing or land rush, but these terms have been criticized for being too biased and suggestive. A key aspect of these terms is that they refer to land purchases by investors, distinguishing them from land trading between private individuals and small-scale farmers. The investors can be domestic or foreign investors (D'Odorico et al., 2017). This is how the term is also used in this paper. Individual land transactions are then referred to as a Large-scale land acquisition (LSLA).

of Afrobarometer. I calculate the period during which the respondents were aged 0 to 16 and determine if they were affected by an LSLA during this time. Since LSLAs are not randomly distributed, I differentiate between active LSLAs, where the land is utilized by investors, and inactive LSLAs, where the land, though purchased, remains unused by investors and thus available to small-scale farmers. This distinction enables me to address the non-random distribution of LSLAs by controlling for the selection bias (Wegenast et al., 2022).

The results from the subsequent OLS regression analysis show that being affected by an active LSLA between the ages of 0 and 16 has a statistically significant negative impact on education. However, the hypothesized stronger negative impact on the education of males is not supported.

This paper makes several contributions to existing research. It contributes to the literature on agricultural shocks while simultaneously broadening the understanding of the socio-economic consequences of LSLAs. Furthermore, it examines the gendered impact of LSLAs, an aspect that has been neglected in the literature, and at the same time, it adds to the body of research that addresses the gendered impacts of agricultural shocks. Additionally, this paper contributes methodologically to the LSLA literature. Most existing LSLA studies are case studies, while geospatial studies are rare, despite their potential to analyze the local consequences of LSLAs.

The paper is structured as follows: Chapter 2 reviews relevant literature and identifies the research gaps this paper aims to address. Chapter 3 outlines the theoretical framework. Chapter 4 describes the research design. Chapter 5 presents the analysis. Chapter 6 focuses on conducting robustness checks and testing the causal mechanisms. Chapter 7 discusses the results, and Chapter 8 concludes with reflections on the findings and suggestions for future research.

2. Literature Review

I divide the literature review into two parts. First, I situate the paper within the literature on agricultural shocks and education, also addressing the gendered impact of agricultural shocks on education. I then provide a brief overview of the literature on LSLAs, as this paper addresses various research gaps in that literature.

3

2.1. Agricultural Shocks and Education

Most of the literature on agricultural shocks and their impact on education deals either with agricultural shocks caused by weather and climate conditions or with agricultural shocks caused by the price decline of agricultural commodities (Alam et al., 2020).

Weather and climate-induced agricultural shocks, such as droughts, floods, or extreme temperature fluctuations, negatively impact the education of those relying on agricultural livelihoods by decreasing agricultural production, which leads to reduced household income and lower educational investment (Miller et al., 2024). Additionally, the income loss often compels families to withdraw children from school to support household finances and mitigate the effects of these shocks (Beegle et al., 2006; Alam et al., 2020). Many small-scale farmers in developing countries are subsistence farmers without a regular income, so the negative impact of weather-induced climate shocks on them comes primarily from the resulting food insecurity (Kinda, 2016; Fusco, 2022). Food insecurity not only directly impairs children's cognitive performance and increases their risk of dropping out of school (Tamiru et al., 2016), but it also compels affected households to withdraw children from school to seek additional income (Bandara et al., 2015). Furthermore, many children are too hungry to attend school, opting instead to search for food (Belachew et al., 2011). The effects of these shocks are profound; for example, Agamile and Lawson (2021) report that rainfall below the local historical average reduces school attendance in Uganda by up to 10%. Similarly, Nübler et al. (2021) observe that lack of rainfall negatively affects both school enrollment and test scores in Kenya, a trend that persists even if the shock occurs before children reach school age, as households struggle to recover from the economic setbacks induced by the shock.

Another research area examines the impact of declining agricultural commodity prices on education. Falling agricultural commodity prices compel farmers to sell their crops for less, which in turn reduces household income (Brown et al., 2023). As financial resources for education decrease, families are forced to pull children from school to contribute to household income through work and cut back on educational investments (Beck et al., 2019). For example, Asfaw (2018) observed that the 2008 global price drop in coffee led to increased dropout rates among children aged 15-18 in Ethiopian coffee-producing villages. Similarly, Cogneu and Jedwab (2012) reported that the 1990 cocoa price drop in the lvory Coast resulted in lower school enrollment rates among children in cocoa-producing households.

4

In developing countries, girls often have less access to education compared to boys, and agricultural shocks can exacerbate these gender disparities (Björkman-Nyqvist, 2013). However, the evidence regarding the differential impact of agricultural shocks on the education of boys and girls remains inconclusive, as outcomes vary based on local conditions and prevailing gender norms. While households may prioritize boys' education, girls often cannot contribute as significantly to household income through labor because of a preference for boys in the labor market (Brown et al., 2023). Research findings highlight cross-regional differences: Beegle et al. (2008) observed that in Tanzania, rainfall shocks negatively affect boys' education more severely. In contrast, Maitra and Tagat (2019) found that in India, girls are predominantly affected by rainfall shocks. Similarly, Björkman-Nyqvist (2013) reported that in Uganda, rainfall shocks disproportionately harm girls' academic performance, as households tend to allocate scarce educational resources preferentially to boys. Lastly, Sen and Villa (2022) discovered that in South Africa, delayed rainfall affects both genders equally.

To the best of the authors' knowledge, no research has examined the impact of LSLAs on education, despite LSLAs constituting a significant agricultural shock by disrupting the agricultural production of small-scale farmers. Additionally, LSLAs impact small-scale farmers as unexpectedly as agricultural shocks induced by weather conditions and declining prices for agricultural commodities. Small-scale farmers are unprepared when production starts on the purchased land, as they usually have no knowledge of the sale of the land because they are working under unclear ownership conditions and are not consulted or informed in advance of the start of production (Vermeulen & Cotula, 2010). Furthermore, studies show that LSLAs lead to income loss (Kebede et al., 2021) and food insecurity (Castet, 2024), the two mechanisms through which agricultural income shocks induced by weather and climate and agricultural income shocks induced by the price decline of agricultural commodities negatively affect education. Given the widespread prevalence of LSLAs in developing countries, it is therefore relevant that the literature on agricultural shocks and education also considers the impact of agricultural shocks induced by LSLAs.

2.2. Determinants and Consequences of LSLAs

The literature can be divided into two main strands: one that explores the determinants of LSLAs and another that examines the consequences of LSLAs for affected communities and small-scale farmers.

Investors typically target areas with favorable soil and climate conditions for agriculture (D'Odorico et al., 2017). They often purchase land where ownership rights are ambiguous and poorly enforced (Yengoh et al., 2016). Land owned by women, who have weaker property rights in many regions, is particularly attractive to investors (Wegenast et al., 2022). Similarly, land belonging to marginalized population groups also attracts LSLAs, as these groups usually lack the political leverage to effectively resist such acquisitions (Moreda, 2017).

Most research on the consequences of LSLAs focuses on the impact of LSLAs on income and food security. LSLAs diminish the food security of individuals living nearby because they lose access to the land they previously cultivated to sustain themselves (Castet, 2024). This also leads to a loss of income for the affected small-scale farmers (Kebede et al., 2021). Despite frequent claims by international actors about job creation through LSLAs, empirical evidence to support this assertion is scant (D'Odorico et al., 2017). Wegenast et al. (2022) note that few studies explore the gendered impacts of LSLAs, even though women, who frequently lack land titles, constitute a significant portion of small-scale farmers. The limited research available suggests that women face greater income losses from LSLAs than men (Yengoh & Armah, 2015) and are forced into more time-consuming tasks like collecting wood and fetching water, which negatively affects their household status (Hajjar et al., 2020).

Some papers find a positive influence of LSLAs on surrounding communities, especially when investors involve locals in agricultural production (Dell'Angelo et al., 2017). Yet, this highlights another issue with the literature on the consequences of LSLAs: it primarily relies on qualitative case studies, whose results are contingent on the specific characteristics of each investment and do not allow for the identification of systematic local relationships (Yang & He, 2021; Wegenast et al., 2022). The infrequent large-N studies often use the aggregated number of LSLAs at the country level as the independent variable (e.g. Brandl et al., 2021), which is inappropriate because the impacts of LSLAs are predominantly local (Hufe & Heuermann, 2017). Geospatial studies are scarce, despite their potential to analyze systematic local effects of LSLAs. Notable exceptions are Wegenast et al. (2022) and Balestri & Maggioni (2019), who show that LSLAs erode family trust and increase conflicts, respectively.

To summarize, this paper makes several contributions to the existing literature. First, it contributes to the literature dealing with the impact of agricultural shocks on education by

6

expanding the focus of this literature to include agricultural shocks induced by LSLAs. Second, it extends the literature on the socio-economic consequences of LSLAs. Furthermore, the paper explores whether LSLAs have a gendered impact on education. Thus, this paper also contributes to the inconclusive literature on the gendered impacts of agricultural shocks. Methodologically, this paper extends beyond the prevalent case studies in LSLA research by using a geospatial approach.

3. Theory

In the following, I present the theoretical explanations regarding the influence of LSLAs on education. I first argue that LSLAs have a negative impact on education through income loss and food insecurity. These mechanisms are interconnected. Income loss can increase food insecurity, but food insecurity alone also undermines education, especially for subsistence farmers. In the second part, I argue that withdrawing children from school to mitigate the adverse effects of LSLAs disproportionately affects boys, resulting in a larger negative impact on their education.

3.1. The Impact of LSLAs on Education

LSLAs lead to income loss for small-scale farmers and surrounding communities. Farmers lose access to the land, preventing them from growing and selling crops (Yengoh & Armah, 2015). Although international investors claim that LSLAs generate employment, evidence often contradicts these assertions, showing that the land is predominantly used for export-oriented and mechanized agriculture, which creates few jobs (Dell'Angelo et al., 2017; Wegenast et al., 2022). The jobs that do arise, such as those in agricultural machinery maintenance, typically require skills not possessed by the local rural population (Sekoai & Yoro, 2016).

Furthermore, LSLAs reduce income even for households not directly dependent on the sold land. The post-acquisition use of land for export-oriented agriculture, biofuel production, or renewable energy creation reduces food availability for local markets and drives up local food prices. This forces households to spend more on food, thereby reducing their income. Müller et al. (2021) found that in areas affected by LSLAs, dietary diversity scores of children under five years old declined, indicating reduced availability of local foods.

7

The loss of income due to LSLAs means that households have fewer financial resources to invest in education. In addition, households withdraw children from school to compensate for income loss through additional work when faced with an agricultural shock (Alam et al., 2020). This occurs even when schooling is free, as there are still substantial associated costs, such as transportation and school materials. For instance, in Africa, school expenses can account for up to 25% of a household's budget (Brown et al., 2023).

Food insecurity is the second mechanism through which LSLAs can negatively affect education. While food insecurity can result from the income loss associated with LSLAs, many small-scale farmers in Africa rely not on regular income, but on consuming the crops they cultivate. The loss of land due to LSLAs prevents them from growing crops for their own consumption, exacerbating food insecurity (Giller et al., 2021). For instance, Kebede et al. (2023) found that individuals affected by LSLAs in Ethiopia had a daily calorie intake 23% lower than that of a control group. Food insecurity impacts education by impairing cognitive performance and increasing dropout risk. Children must also spend time searching for food, reducing time spent on education (Tamiru et al., 2016). Furthermore, like income loss, food insecurity can lead households to withdraw children from school to help generate income and mitigate food shortages (Belachew et al. 2011). My arguments lead to the following hypothesis:

H1: LSLAs negatively impact the education of people living in their vicinity.

3.2. The Gendered Impact of LSLAs on Education

As outlined in the literature review, the evidence regarding the gendered impact of agricultural shocks on education is inconclusive. However, research indicates that when a community loses access to land due to an LSLA, men are more likely to secure jobs in rural areas (Hajjar et al., 2020). Conversely, women struggle to find alternative employment opportunities. They face limited options outside agriculture, as men are often preferred for physical labor, and many non-agricultural jobs also require hard physical work (Yengoh & Armah, 2015).

I argue that the fact that women often have fewer employment opportunities than men in rural areas means that the negative impact of LSLAs on education is more pronounced for men because men are more able to contribute to household income through work and thus can no longer attend school. Given the substantial agricultural shock resulting from LSLAs, I contend that the pressure imposed by LSLAs—through income loss and heightened food insecurity— leads households to disregard existing gender-specific educational preferences. Supporting this, research indicates that in Sub-Saharan Africa, where many LSLAs occur, households abandon these preferences when faced with substantial income shocks, with families frequently withdrawing boys from school due to their greater potential to contribute to household income (Brown et al., 2023). An aspect that reinforces my theory is that agricultural shocks induced by LSLAs are irrecoverable for small-scale farmers in the long term, unlike other agricultural shocks, because they result in permanent loss of land access. Therefore, farmers cannot simply endure the consequences of LSLAs while maintaining their gender-specific preferences; instead, they must immediately and permanently seek new sources of income. This leads to the following hypothesis.

H2: The negative impact of LSLAs on education is more pronounced for males.

4. Research Design

I analyze the two hypotheses, depicted in Figure 1, in the African context for three reasons. First, education remains a critical challenge for African governments; only 70% of African children finish primary school, and just 30% complete lower secondary school (Brown et al., 2023). Second, Africa is a major target for LSLAs. Between 2005 and 2020, land purchased by investors in Africa increased from 1 million to 7.3 million hectares, with 56% of this land previously used by small-scale farmers and pastoralists, securing livelihoods for much of the rural population (Wegenast et al., 2022). Third, the agricultural sector is the most important employment sector in Africa, providing the basis for income and food security for significant parts of the population (Davis et al., 2023).





I employ a geospatial research design to test the hypotheses, using data from 46,711 respondents from the sixth round of Afrobarometer and 322 LSLAs from the Land Matrix database. The analysis is conducted at the individual respondent level.

The sixth round of the Afrobarometer is the most recent geocoded round. Afrobarometer uses a clustered, stratified, multi-stage area probability sampling design to create a representative cross-section of voting-age citizens in each country (Afrobarometer, 2024). Respondents are interviewed within enumeration areas, the smallest geographical units with population data, which are geocoded to assign coordinates to each respondent (BenYishay et al., 2017).

The Land Matrix³ database contains geodata on LSLAs worldwide, primarily derived from public sources such as newspapers and government reports. It includes LSLAs involving the transfer of land rights through sale, lease, or concession, typically covering at least 200 hectares. In some countries, such as Uganda, Senegal, and Cameroon, LSLAs covering only 50 hectares are included (Land Matrix, 2024).

I start by discussing the identification strategy and data selection together, as the identification strategy is crucial for comprehending the data selection and the subsequent discussions on the operationalization of the variables and model specification.

³ The Land Matrix database is a joint project of the German Corporation for International Cooperation (GIZ), the Centre for Development and Environment of the University of Bern, the German Institute for Global and Area Studies and the International Land Coalition (Land Matrix, 2024).

4.1. Identification Strategy and Data Selection

I calculated for each respondent the period the respondent was between 0 and 16 years old (0 = Survey Year - Respondent Age; 16 = 0 + 16), as LSLAs can only negatively affect education if respondents had not completed their education at the time of the land purchase. I use 0 as the lower bound because agricultural shocks at birth negatively affect education, with households taking a long time to recover financially and consequently not sending children to school initially (Nübler et al., 2021). I use 16 as the upper bound due to the structure of my data. The Afrobarometer surveys were conducted in 2014 and 2015. Therefore, I only include LSLAs where the investor purchased the land by 2013. If I used 18 as the upper bound, the 0-18 year period for some younger respondents would extend beyond 2013, where I no longer consider LSLAs. This would result in a shorter effective time period for younger respondents compared to older respondents whose periods do not extend beyond 2013. Reducing the timeframe is not possible, as it would result in too few respondents being affected by LSLAs. The negative impact of agricultural shocks on education are persistent (Nübler et al., 2021). Therefore, the negative impact of LSLAs on education, when respondents were between 0 and 16 years old, can be observed in their education levels at the time of the Afrobarometer survey.

Next, I converted both the Afrobarometer and Land Matrix data sets into *simple feature objects* using the *simple features* R package. Then, I harmonized the coordinate systems of both data sets using the Mercator coordinate system with the identifier 32732. Afterwards, I created buffers around the LSLAs.

Wegenast et al. (2022) use a 10-kilometer buffer in their geospatial analysis of the impact of LSLAs on trust. I decided to use a 25-kilometer buffer instead for five reasons⁴. First, the Afrobarometer data does not allow for controlling for migration movements, meaning it is unclear whether respondents moved after their schooling or attended school near an LSLA and subsequently relocated. A 25-kilometer buffer helps to address this issue by covering a broader area. Second, LSLAs often cover large areas. The average size of the LSLAs I considered is 27,135.67 hectares, with the largest LSLA being 168,000 hectares. Third, I argued that LSLAs

⁴ A 25-kilometer buffer is still rather restrictive. Most researchers use a buffer size of 50 kilometers in geospatial analyses (Wegenast et al., 2022).

lead to income loss through rising local market prices. If a LSLA prevents small-scale farmers in a region from growing crops for local markets, is likely to extend beyond 25 kilometers. Fourth, geocoded data often contain location errors, and larger buffers are more robust against such errors (Knutsen et al., 2017). Lastly, a smaller buffer size would result in too few respondents being affected.

Investors consider various factors when purchasing land, resulting in a non-random distribution of LSLAs in Africa (Wegenast et al., 2022). Since LSLAs may be concentrated in areas with below or above-average education levels, comparing the education of individuals living near LSLAs with those who do not is problematic due to potential selection bias.



Figure 2: Identification Strategy

Note: Created by the author.

To address the selection bias, I modify the identification strategy by Wegenast et al. (2022)⁵. My approach is illustrated in Figure 2. I classify LSLAs into two types: active LSLAs, where the investor actively uses the land, and inactive LSLAs, where the land is purchased by the investor but not used. The assumption behind this is that LSLAs negatively impact the local population only when the land is used by the investors, not when it is purchased but left unused. Consequently, I assume that small-scale farmers retain access to the land until it begins to be utilized. This assumption is justified for two reasons. First, many small-scale farmers already cultivate land with unclear ownership before the LSLA, often remaining unaware of the purchase until production starts (Yengoh et al., 2016). Second, many investors buy land for

⁵ My identification strategy builds on the approach developed by Knutsen et al. (2017) and adapted by Wegenast et al. (2022) for LSLAs. I refined it by considering both the age of respondents and their positions within buffers, unlike the other approaches which only consider respondent positions. Moreover, unlike Wegenast et al. (2022), I created buffers around the LSLAs instead of the respondents, as I find this more straightforward. The outcome is the same.

speculation (Dell'Angelo et al., 2017) and have no interest in evicting small-scale farmers, as expulsion incurs costs (Sändig, 2021).

I proceed by dividing the Afrobarometer respondents into three groups: 1) Those within the buffer of an active LSLA and aged 0-16 when land use began, 2) Those within the buffer of an inactive LSLA and aged 0-16 at the time of land purchase, and 3) Those unaffected by any LSLAs during ages 0-16. In Figure 2, one respondent, represented by a black triangle, is within the buffer of an active LSLA and was between 0 and 16 years old when the investor began using the land; hence, he or she belongs to the first group. Another respondent, shown as a blue triangle, is in the same buffer but was 60 years old at the onset of land use, thus is categorized in the reference group, like other respondents outside any LSLA buffer. The same logic applies to the buffer and aged 0-16 at the time of purchase, and therefore belong to the second group. The other respondent in the buffer was over 16 at the time of the land purchase and therefore belongs to the third group.

In the subsequent regression analysis, the coefficient of respondents in the buffers of active LSLAs during ages 0 to 16 (β_1) then includes both the selection bias and the impact of LSLAs on education while the coefficient of respondents in the buffer of inactive LSLAs during ages 0 to 16 (β_2) only includes the selection bias. Taking the difference between the coefficients of active and inactive LSLAs ($\beta_1 - \beta_2$) removes the selection bias. Subsequently, an F-test is conducted to ensure that the difference between the coefficients is statistically significant. This approach is feasible because active and inactive LSLAs are largely similar⁶. The difference is that investors with inactive LSLAs speculate with the land without starting cultivation or are still preparing for production (Wegenast et al., 2022).

Now, I outline the data selection process. Initially, the sixth round of the Afrobarometer includes 52,567 respondents from 36 countries (Afrobarometer, 2016). I removed respondents who aged 0 to 16, were within a 25-kilometer buffer of both inactive and active LSLAs, as it is impossible to determine which LSLA influenced their education. I excluded all respondents from Algeria, Burundi, Cape Verde, Niger, and Mauritius because the Land Matrix

⁶ In my case, the median hectare size of active LSLAs is 6774 hectares, while for inactive LSLAs, it is 6474 hectares.

database records no LSLAs in these countries. This accounts for unobserved factors that may make a country unattractive for LSLAs and could influence the characteristics of the group not affected by LSLAs. My final sample includes 46,711 respondents from 31 African countries. 2.36% of the respondents are affected by an active LSLA between the ages of 0 and 16, and 0.24% are affected by an inactive LSLA during this period⁷. Figure 3 displays the locations of the respondents considered.





Note: Created by the author.

For my analysis, I only include LSLAs with precise coordinates. Despite this, location errors are possible. I consider all LSLAs where the purchase has been completed, regardless of whether the land was bought by a domestic or international investor and irrespective of the post-purchase form of land utilization⁸. I define them as active if the land is used by the investor and inactive if it is not. LSLAs with unclear land use status are excluded. I use the purchase year as the LSLA year since land use typically begins immediately for active LSLAs. If the year

⁷ The figures are comparable to Wegenast et al. (2022). Using a 50 km buffer as a robustness check, 6.46% of respondents are affected by active LSLAs (0-16 years old) and 0.76% by inactive LSLAs. These values are higher than those of Wegenast et al. (2022) and the results remain robust (Table C.1).

⁸ I do not differentiate because my sample of LSLAs would be too small, resulting in an insufficient number of respondents affected by LSLAs. Forms of post-purchase land utilization include export-oriented agriculture, biofuel production, and renewable energy production (Land Matrix, 2024)

is missing, I manually check the Land Matrix database for the relevant year. LSLAs without any information on the completion year are excluded. Since the sixth round of Afrobarometer surveys began in 2014, I only include LSLAs purchased before 2013. My final sample includes 322 LSLAs in Africa from 1921 to 2013, with 277 being active and 45 inactive. Figure 4 displays the locations of the LSLAs included in the analysis⁹.



Figure 4: Locations of LSLAs

Note: Created by the author.

4.2. Operationalization

The dependent variable is *education*, which is measured in the sixth round of Afrobarometer by asking respondents, "What is your highest level of education?" Respondents indicated their education level on a Likert scale, where higher values correspond to higher levels of education. Specifically, 0 indicates no formal schooling, 1 signifies informal schooling (including Koranic schooling), 2 denotes some primary schooling, 3 means primary school completed, 4 represents intermediate or some secondary/high school, 5 indicates secondary/high school completion, 6 refers to post-secondary qualifications other than a university degree (e.g., a diploma from a polytechnic or college), 7 indicates some university education, 8 implies

⁹ Appendix A contains a list of all Afrobarometer and Land Matrix countries considered.

university completion, and 9 stands for post-graduate education (Afrobarometer, 2016)¹⁰. Figure 5 illustrates that the majority of respondents have an education level below secondary education completion.



Figure 5: Histogramm of Education Levels

Note: Created by the author.

The independent variable is *active*, indicating whether an individual is affected by an active LSLA between the ages of 0 and 16. *Active* is coded as 1 if an individual is within the buffer of an active LSLA during this age range when land utilization began, and 0 if not.

To investigate the gendered impact of LSLAs on education, I use an interaction term between *active* and *gender*. *Gender* is a dummy variable, with 1 for male respondents and 0 for female respondents.

I control for various confounders. Most variables in the Afrobarometer dataset are posttreatment, meaning their values when respondents were between 0 and 16 years old are unknown. Therefore, I use variables that represent permanent characteristics and do not

¹⁰ I address potential issues with measuring *education* through a Likert scale in Chapter 6.

change over time. Additionally, I utilize control variables from grid data, calculating values for each respondent during the period they were between 0 and 16 years old. Each respondent is matched to their respective grid based on coordinates.

To address the selection bias, I use the variable *inactive*, indicating whether a respondent was affected by an inactive LSLA between the ages of 0 and 16. It takes the value 1 if a respondent is within the buffer of an inactive LSLA during this age range at the time of land purchase, and 0 if not.

To account for differences in the duration respondents are affected by LSLAs, I use the variable *exposure time*. It indicates the number of years between ages 0 and 16 that a respondent is affected by an active or inactive LSLA. For the reference category, this value is always 0. This variable is particularly useful as it also captures the age at which each respondent was affected by the LSLA. For instance, if the variable takes the value 5, it means the respondent was 11 years old when the land was purchased or when the utilization of the purchased land began.

I use gender, religion, and age dummy variables as control variables from the Afrobarometer dataset. Men generally have better access to education and are more likely to hold jobs created by LSLAs, making them more likely to live near LSLA locations (Björkman-Nyqvist, 2013; Yengoh & Armah, 2015). *Gender* is operationalized as in the interaction term. *Religion* is coded as 1 if the respondent belongs to any Muslim, Sunni, Ismaeli, Mouridiya, Tijaniya, Qadiriya, Shia, or traditional/ethnic religion; otherwise, it is 0. These religious groups often have poorer access to education, and women in these groups frequently face educational barriers (Alesina et al., 2023). Age dummies are used to account for differences in access to education between younger and older individuals, and to control for the timing of the period when respondents were between 0 to 16 years old, as access to education has improved over time (Brown et al., 2023). Each respondent is coded with a 1 for the age group they belong to and 0 for all other groups. The age groups are 18-25, 26-35, 36-45, 46-55, 56-65, and 66+, with 46-55 serving as the reference category.

Respondents in areas with good agricultural conditions are more likely to be affected by LSLAs (D'Odorico et al., 2017) and potentially have higher incomes, allowing for greater educational investment. To control for agricultural conditions, I use two variables. First, I utilize data from the gridded dataset by Schneider et al. (2015), which provides yearly precipitation

measurements in millimeters for each grid, derived from monthly meteorological statistics compiled by the Global Precipitation Climatology from 1946 to 2013. To operationalize *rain*, I calculated the average precipitation in each respondent's grid for the period when the respondent was between 0 and 16 years old. Additionally, I control for the agricultural land available to the respondent and LSLA investors using the dataset by Meiyappan et al. (2012). This dataset reports the percentage of a grid's area covered by agricultural land from 1950 to 2010. To operationalize *agricultural land*, I calculated the average percentage of agricultural land in each respondent's grid for the period when the respondent was between 0 and 16

To control for other agricultural shocks that might negatively impact a respondent's education, I utilize drought data from Beguería et al. (2010). They measure drought intensity by calculating, for each grid and year, the proportion of months within the growing season with values below -1.5 in the Standardized Precipitation and Evapotranspiration Index (SPEI-1). The growing season is defined as the months when the main crop of each grid is cultivated, with values below -1.5 indicating severe drought conditions. To operationalize *drought*, I calculated the average number of months with SPEI-1 values below -1.5 during the growing season for each respondent's grid when they were between 0 and 16 years old.

Access to education is often better in urban areas (Björkman-Nyqvist, 2013). To account for this, I use the control variable *urban*, which I operationalize by calculating the average percentage of each respondent's grid covered by urban areas in the time the respondent was between 0 and 16 years old. The data for this come from Meiyappan et al. (2022). Descriptive statistics of the variables are shown in Table 1.¹².

¹¹ I sourced the descriptions of the datasets for the grid-based variables from the PRIO-GRID Codebook, citing them as instructed (see Tollefsen et al., 2012).

¹² Missing observations for both Afrobarometer variables and grid-based control variables were coded as NAs. The number of observations considered for calculating grid-based variables may vary from respondent to respondent.

Variable	Ν	Mean	SD	Min	Max
Education	46710	3.478	2.197	0	9
Active	46711	0.024	0.152	0	1
Inactive	46711	0.002	0.049	0	1
Gender	46710	0.495	0.5	0	1
Religion	46711	0.29	0.454	0	1
Urban	44868	0.563	1.192	0	19.865
Drought	43456	0.05	0.025	0	0.6
Rain	46294	588.667	303.049	0.321	1082.364
Agricultural Land	44868	33.196	16.035	0	99.03
Exposure Time	46468	0.103	0.942	0	16
Age (18 -25)	46431	0.245	0.43	0	1
Age (26-35)	46431	0.304	0.46	0	1
Age (36-45)	46431	0.204	0.403	0	1
Age (46-55)	46431	0.125	0.331	0	1
Age (56-65)	46431	0.074	0.262	0	1
Age (66 +)	46431	0.049	0.216	0	1

Table 1: Descriptive Statistics

4.3. Model Specification

I use model 1 to test my first hypothesis. For hypothesis 2, I employ two distinct approaches: First, model 2 incorporates an interaction term between *active* and *gender*. Second, following Wegenast et al. (2022), I apply model 1 without using *gender* as a control variable on gendersegregated datasets. This allows me to remove the selection bias, which is not possible with the interaction term. I employ OLS regressions with region fixed effects to address timeinvariant regional characteristics that may affect respondent education. Additionally, I utilize heteroskedasticity-robust standard errors. Below, my regression models are presented.

Model 1: $Education_{T(S)ilr} = \beta_0 + \beta_1 Active_{T(0-16)ilr} + \beta_2 Inactive_{T(0-16)ilr} + X_{T(S)ilr}\theta_X + Z_{T(0-16)ilr}\theta_Z + rf_r + \varepsilon_{T(S)ilr}$

In model 1, $Education_{T(S)ilr}$ represents the education level of individual *i* in locality *l*, region r and the time of the survey T(S). $\beta_1 Active_{T(0-16)ilr}$ indicates whether individual *i* is within the buffer of an active LSLA in locality *l* and region r within the time period T(0 - 16), when the respondent was between 0 and 16 years old. $\beta_2 Inactive_{T(0-16)ilr}$ indicates whether

individual *i* is within the buffer zone of an inactive LSLA in locality *l* and region *r* within the time period T(0-16). $X_{T(S)ilr}$ is a vector of all constant individual-level control variables measured at the time of each survey, and θ_X is the vector of corresponding coefficients. $Z_{T(0-16)ilr}$ is a vector of all grid-based control variables specific to the period when the respondent was aged between 0 and 16, and θ_Z contains the corresponding coefficients. rf_r represents the region fixed effects, and $\varepsilon_{T(S)ilr}$ is the error term.

$$Model 2: Education_{T(S)ilr} = \beta_0 + \beta_1 Active_{T(0-16)ilr} + \beta_2 Inactive_{T(0-16)ilr} + \beta_3 Gender_{T(S)ilr} + \beta_4 \left(Active_{T(0-16)lr} x \ Gender_{ilr} \right) + X_{T(S)ilr} \theta_X + Z_{T(0-16)ilr} \theta_Z + rf_r + \varepsilon_{T(S)ilr}$$

Model 2 differs from Model 1 insofar as I have included the interaction term $\beta_4 \left(Active_{T(0-16)ilr} x \ Gender_{T(S)ilr} \right)$.

5. Results

I focus initially on testing hypothesis 1 and then proceed to hypothesis 2. I do not reject hypothesis 1 if the difference between *active* and *inactive* ($\beta_1 - \beta_2$) results in a negative value, both coefficients are statistically significant, and the difference between the two coefficients is statistically significant. Hypothesis 2 is supported if the interaction term is statistically significant and goes in the expected direction, and the difference between *active* and *inactive* yields a larger negative value for men in the gender-differentiated data sets. As with hypothesis 1, both the individual coefficients and their difference must be statistically significant.

Figure 6 shows the results of the regression analysis. The coefficients of *active* and *inactive* are statistically significant (p < 0.05) and the difference between the two coefficients yields a negative value of -0.36. To test whether the difference between *active* and *inactive* is statistically significant, I conducted an F-test to examine whether I can reject the null hypothesis that *active* and *inactive* are similar ($\beta_1 - \beta_2 = 0$). The result (F = 2.71, p < 0.1) indicates that the difference is statistically significant (see Regression Table in Appendix B.1). Thus, I can confirm hypothesis 1.





Note: * *p* < 0.05, ** *p* < 0.01, *** *p* < 0.001. Robust standard errors are clustered by region. Negative coefficients are in red, positive coefficients in blue, with 95% confidence intervals shown. Regional fixed effects are applied. The reference group for active and inactive is respondents not affected by any LSLA during ages 0 to 16. For other dummy variables, the reference group is those with a value of 0. The reference group for age dummies is 46-55. Regressions were estimated using the Fixest package in *R*, which does not report an intercept by default.

Regarding the control variables, it is noteworthy that *drought* has a substantive statistically significant negative impact on education (p < 0.05), aligning with the existing literature. Additionally, *exposure time* has a statistically significant negative impact on education (p < 0.05). *Gender* (p < 0.001), *religion* (p < 0.001), *urban* (p < 0.001), and the age dummies (p < 0.001) also have statistically significant effects on education in the expected direction.

I now present the results of the analysis on the gendered impacts of LSLAs on education. The interaction term between *active* and *gender* is not statistically significant (see Regression Table in Appendix B.2.). Figure 7 illustrates the marginal effects of the interaction. The predicted education value for males does not decrease, as expected, when *active* switches from 0 to 1.



Figure 7: Marginal Effects of Interaction Term

The interaction term does not account for selection bias, so I separated the dataset by gender to examine whether the difference between *active* and *inactive* yields a larger negative value for the male dataset.

I checked whether being affected by an LSLA is influenced by gender. 2.72% of women and 2.48% of men are affected by an inactive or active LSLA during the ages 0 to 16. Despite the small difference, the results should be interpreted with caution.

Figure 8 shows that, contrary to my expectations, active LSLAs have a larger negative impact on females (see Regression Table in Appendix B. 3). The difference between coefficients is not statistically significant in either dataset.

Note: All other variables are held constant at their means. The bars around the coefficients represent the 95% confidence intervals.

Figure 8: Regression Analysis (Gender Differentiated)



Note: * p < 0.05, ** p < 0.01, *** p < 0.001. The models include all control variables except gender. The coefficients are presented with 95% confidence intervals. The model includes regional fixed effects and robust standard errors clustered by region.

Since the interaction term is not statistically significant, and the difference between active and inactive does not yield a larger negative value for males compared to females, I reject hypothesis 2.

6. Robustness Checks and Causal Mechanisms Testing

I conducted various robustness checks (see Appendix C) and subsequently highlight a few key findings. I tested hypothesis 1 using a buffer size of 50 Kilometers (Table C.1). Active (p < 0.1) and *inactive* (p < 0.001) are statistically significant. The difference yields a negative value of – 0.38 and is statistically significant (F = 9.04, p < 0.01). Moreover, I replaced *religion* with *ethnicity* to address that religion, in some cases, might be a post-treatment variable too. My results remain robust (Table C.10). Both *active* (p < 0.1) and *inactive* (p < 0.01) are statistically

significant. Taking the difference yields a negative value of -0.4 and the difference is statistically significant (F = 3.39, p < 0.1).

Additionally, I operationalized the dependent variable *education* by creating binary variables for different educational stages and conducted logit regressions. This approach addresses the issue of running OLS regressions on ordered outcomes, which could violate the assumption of independent error terms (Wegenast et al., 2022), and accounts for respondents under 30 who may not have completed their education. The coefficient differences usually trend as expected: LSLAs increase the likelihood of having only primary education and decrease the likelihood of having secondary education. However, the coefficients are not statistically significant in most models, and their differences are not significant in any of the models (Tables C.4 & C.7).

The interaction term is not significant in almost all logit models and does not follow the expected direction. Although *active* and *inactive* are not statistically significant in most models, the analysis based on gender-separated datasets suggests that LSLAs have a stronger negative impact on males' education (Tables C.2,C.3,C.5,C.6,C.8,C.9,C.11,C.12).

I also tested my theory by examining four conditions: LSLAs increase food insecurity, reduce income, that reduced income and food insecurity negatively impact education, and that LSLAs do not have a negative impact on school infrastructure. The results are mixed. LSLAs improve school infrastructure, allowing me to rule out that the negative impact on education is due to a deterioration of the education supply side. However, LSLAs do not have a statistically significant effect on income or food insecurity, although the coefficient difference points in the expected direction. Nonetheless, the impact of food insecurity and income loss on education is statistically significant and negative (see Appendix D for analysis and discussion).

7. Discussion

The analysis indicates that LSLAs have a negative effect on education, with no evidence that this effect is more pronounced for men. The results remain robust when increasing the buffer size and the negative impact of LSLAs on education is not due to a deterioration of school infrastructure. However, the F-test in main the analysis is only significant at the 10% level and the difference between *active* and *inactive is* no longer significant when using logit models. Given the mixed results from robustness checks and causal mechanisms testing, the findings

24

should be interpreted with caution. Additionally, there are several limitations to my analysis, which I will outline next.

My research design does not control for migration movements, implying I cannot determine if respondents lived near the LSLA during the period when they were between 0 and 16 years old or if they moved there later. Although most respondents are young, so the time when they were between 0 and 16 is relatively close to the survey date, and I used a 25-kilometer buffer to account for migration, it remains a significant limitation.

Using grid-based variables is another limitation, as local conditions within grids can vary. However, grid-based variables are more suitable than other Afrobarometer control variables, which are almost certainly post-treatment variables.

Less than a third of the LSLAs I analyzed are inactive, yet in reality, inactive LSLAs greatly outnumber active ones as land speculation is a major incentive for investors (Dell'Angelo et al., 2017). Therefore, the proportion of people affected by inactive LSLAs is likely underestimated in my analysis. Capturing inactive LSLAs is challenging because investors often avoid public scrutiny (Land Matrix, 2024).

Using the Afrobarometer dataset also has limitations. Only 36 African countries were included in the sixth round, with disproportionate representation of West and East Africa, limiting the external validity of my results. External validity is a broader issue in LSLA literature. Africa represents an extreme case. While LSLAs also have significant negative impacts on communities in Latin America and Asia (Dell'Angelo et al., 2021), a smaller proportion of people in these regions depend on agriculture (Davis et al., 2023). Additionally, investors buy more land there, which was already used for export-oriented agriculture compared to Africa (Müller et al., 2021). Moreover, the external validity of theories, including mine, regarding the gendered impact of LSLAs is constrained by significant variations in gender perceptions across and within regions, making these theories rather educated guesses than definitive conclusions.

My analysis highlights a terminological issue in the LSLA literature. Wegenast et al. (2022) and I have investigated the impact of actively used LSLAs on the local population. This means that the impact on the local population is not due to the land purchase itself, but rather the utilization of the land after the purchase. This distinction needs to be made clear in the literature. Additionally, this leads to a controversial hypothesis that warrants further investigation: land speculation is not as problematic as land utilization because farmers do not lose access to the land.

Despite its limitations, this paper contributes to the literature on agricultural shocks and the socioeconomic effects of LSLAs. It argues for recognizing LSLAs as agricultural shocks, expanding the focus beyond those caused by weather or market price fluctuations. It also introduces education as a critical socioeconomic indicator affected by LSLAs, which has not been previously examined.

The findings have significant policy implications. LSLAs, often promoted by African governments as catalysts for agricultural modernization and development in impoverished rural areas (Wegenast et al., 2022), negatively impact education. This can trap individuals in poverty by leaving them without the qualifications needed for modern agribusiness jobs, further impoverishing rural regions. Governments should ensure sustainable land use by investors. Policies requiring the employment of small-scale farmers who previously cultivated the land and providing these farmers with official land titles would protect them from unexpected LSLAs and improve their negotiation positions with investors.

8. Conclusion

The aim of this paper was to answer the research question: What is the impact of LSLAs on education? I argued that LSLAs negatively affect education by increasing food insecurity and reducing income, leaving households with fewer resources for education and forcing them to withdraw children from school to generate additional income. I also argued that the negative effect is more pronounced for boys, as they can contribute more to household income and are thus more likely to be taken out of school.

To test the hypotheses, I used a geospatial research design. The regression results indicate that being affected by LSLAs between the ages of 0 and 16 has a statistically significant negative impact on education. I found no evidence that this negative effect is stronger for boys.

26

There are many avenues for future research, such as differentiating LSLAs by international and domestic investors and different land uses. LSLAs have far-reaching environmental impacts, including soil degradation and reduced biodiversity (D'Odorico et al., 2017). This potentially makes smallholders more vulnerable to other agricultural shocks. Future research should therefore investigate how LSLAs and other agricultural shocks interact. Moreover, integrating institutional variables into research on agricultural shocks and education is promising, as the impact of these shocks on education may be moderated by factors such as local institutional quality. Research on LSLAs should expand beyond Africa. Since the proportion of people dependent on agricultural income is lower in Asia and Latin America, it is important to focus on vulnerable groups, particularly indigenous populations, in these regions.

9. References

- Afrobarometer. (2016). Merged round 6 codebook (36 countries). Retrieved January 8, 2024, from https://www.afrobarometer.org/survey-resource/merged-round-6-codebook-36-countries-2016/
- Afrobarometer. (2024). Sampling principles and weighting. Retrieved January 5, 2024, from https://www.afrobarometer.org/surveys-and-methods/sampling/
- Agamile, P., & Lawson, D. (2021). Rainfall shocks and children's school attendance: Evidence from Uganda. *Oxford Development Studies*, *49*(3), 291–309. https://doi.org/10.1080/13600818.2021.1895979
- Alam, S. A., Pörtner, C. C., & Simpson, C. (2020). Economic Shocks and Children's Education.
 Handbook of Labor, Human Resources and Population Economics (pp. 1– 19).https://doi.org/10.1007/978-3-319-57365-6_311-1
- Alesina, A., Hohmann, S., Michalopoulos, S., & Papaioannou, E. (2023). Religion and educational mobility in Africa. *Nature*, *618*(7963), 134–143. https://doi.org/10.1038/s41586-023-06051-2
- Asfaw, A. A. (2018). The effect of coffee price shock on school dropout: New evidence from the 2008 global financial crisis. *Applied Economics Letters*, *25*(7), 482–486. https://doi.org/10.1080/13504851.2017.1340560
- Balestri, S., & Maggioni, M. A. (2021). This Land Is My Land! Large-Scale Land Acquisitions and Conflict Events in Sub-Saharan Africa. *Defence and Peace Economics*, 32(4), 427– 450. https://doi.org/10.1080/10242694.2019.1647727
- Bandara, A., Dehejia, R., & Lavie-Rouse, S. (2015). The Impact of Income and Non-Income Shocks on Child Labor: Evidence from a Panel Survey of Tanzania. World Development, 67, 218–237. https://doi.org/10.1016/j.worlddev.2014.10.019
- Beck, U., Singhal, S., & Tarp, F. (2019). Commodity Prices and Intra-Household Labor Allocation. American Journal of Agricultural Economics, 101(2), 436–454. https://doi.org/10.1093/ajae/aay082
- Beegle, K., Dehejia, R. H., & Gatti, R. (2006). Child labor and agricultural shocks. Journal of Development Economics, 81(1), 80–96. https://doi.org/10.1016/j.jdeveco.2005.05.003
- Beegle, K., Dehejia, R. H., Gatti, R., & Krutikova, S. (2008). *The Consequences of Child Labor: Evidence from Longitudinal Data in Rural Tanzania.*
- Beguería, S., Vicente-Serrano, S. M., & Angulo-Martínez, M. (2010). A Multiscalar Global Drought Dataset: The SPEIbase: A New Gridded Product for the Analysis of Drought Variability and Impacts. *Bulletin of the American Meteorological Society*, *91*(10), 1351–1356. https://doi.org/10.1175/2010bams2988.1
- Belachew, T., Hadley, C., Lindstrom, D., Gebremariam, A., Lachat, C., & Kolsteren, P. (2011).
 Food insecurity, school absenteeism and educational attainment of adolescents in
 Jimma Zone Southwest Ethiopia: A longitudinal study. *Nutrition Journal*, 10(1), 29.
 https://doi.org/10.1186/1475-2891-10-29

BenYishay, A., Rotberg, R., Wells, J., Lv, Z., Goodman, S., Kovacevic, L., & Runfola, D. (2017). Geocoding afrobarometer rounds 1-6: Methodology & data quality.

- Björkman-Nyqvist, M. (2013). Income shocks and gender gaps in education: Evidence from Uganda. *Journal of Development Economics*, *105*, 237–253. https://doi.org/10.1016/j.jdeveco.2013.07.013
- BMZ (Bundesministerium f
 ür wirtschaftliche Zusammenarbeit und Entwicklung). (2024). Education in developing countries. Retrieved March 15, 2024, from https://www.bmz.de/en/issues/education-a-human-right/education-in-developingcountries-197598
- Bömmel, N., & Heineck, G. (2023). Revisiting the causal effect of education on political participation and interest. *Education Economics*, 31(6), 664–682. https://doi.org/10.1080/09645292.2022.2141199
- Brandl, K., Moore, E., Meyer, C., & Doh, J. (2022). The impact of multinational enterprises on community informal institutions and rural poverty. *Journal of International Business Studies*, 53(6), 1133–1152. https://doi.org/10.1057/s41267-020-00400-3
- Brown, K. J., Haer, R., & Østby, G. (2023). Local food price volatility and school dropout in Sub-Saharan Africa. *Population and Development Review*, *49*(3), 443-468
- Castet, A. (2024). The impact of large-scale land acquisitions on child food insecurity in Africa. *World Development*, *179*, 106597. https://doi.org/10.1016/j.worlddev.2024.106597
- Cogneau, D., & Jedwab, R. (2012). Commodity Price Shocks and Child Outcomes: The 1990 Cocoa Crisis in Côte d'Ivoire. *Economic Development and Cultural Change*, *60*(3), 507– 534. https://doi.org/10.1086/664017
- Davies, N. M., Dickson, M., Davey Smith, G., Windmeijer, F., & van den Berg, G. J. (2023). The causal effects of education on adult health, mortality and income: Evidence from Mendelian randomization and the raising of the school leaving age. *International Journal of Epidemiology*, 52(6), 1878–1886. https://doi.org/10.1093/ije/dyad104
- Davis, B., Mane, E., Gurbuzer, L.Y., Caivano, G., Piedrahita, N., Schneider, K., Azhar, N.,
 Benali, M., Chaudhary, N., Rivera, R., Ambikapathi, R., Winters, P. (2023). *Estimating* global and country-level employment in agrifood systems. Food & Agriculture Org.
- Dell'Angelo, J., D'Odorico, P., Rulli, M. C., & Marchand, P. (2017). The Tragedy of the Grabbed Commons: Coercion and Dispossession in the Global Land Rush. *World Development*, 92, 1–12. https://doi.org/10.1016/j.worlddev.2016.11.005
- Dell'Angelo, J., Navas, G., Witteman, M., D'Alisa, G., Scheidel, A., & Temper, L. (2021).
 Commons grabbing and agribusiness: Violence, resistance and social mobilization.
 Ecological Economics, 184, 107004. https://doi.org/10.1016/j.ecolecon.2021.107004
- D'Odorico, P., Rulli, M. C., Dell'Angelo, J., & Davis, K. F. (2017). New frontiers of land and water commodification: Socio-environmental controversies of large-scale land acquisitions. *Land Degradation & Development, 28*(7), 2234–2244. https://doi.org/10.1002/ldr.2750
- Farias, D. B. L. (2023). Country differentiation in the global environmental context: Who is 'developing' and according to what? *International Environmental Agreements:*

Politics, Law and Economics, 23(3), 253–269. https://doi.org/10.1007/s10784-023-09596-9

- Fusco, G. (2022). Climate Change and Food Security in the Northern and Eastern African Regions: A Panel Data Analysis. *Sustainability*, 14(19), 12664. https://doi.org/10.3390/su141912664
- Giller, K. E., Delaune, T., Silva, J. V., van Wijk, M., Hammond, J., Descheemaeker, K., van de Ven, G., Schut, A. G. T., Taulya, G., Chikowo, R., & Andersson, J. A. (2021). Small farms and development in sub-Saharan Africa: Farming for food, for income or for lack of better options? *Food Security*, *13*(6), 1431–1454. https://doi.org/10.1007/s12571-021-01209-0
- Hajjar, R., Ayana, A. N., Rutt, R., Hinde, O., Liao, C., Keene, S., Bandiaky-Badji, S., &
 Agrawal, A. (2020). Capital, labor, and gender: The consequences of large-scale land transactions on household labor allocation. *The Journal of Peasant Studies*, 47(3), 566–588. https://doi.org/10.1080/03066150.2019.1602520
- Hofmarcher, T. (2021). The effect of education on poverty: A European perspective. *Economics of Education Review*, *83*, 102124. https://doi.org/10.1016/j.econedurev.2021.102124
- Hufe, P., & Heuermann, D. F. (2017). The local impacts of large-scale land acquisitions: A review of case study evidence from Sub-Saharan Africa. *Journal of Contemporary African Studies*, *35*(2), 168–189. https://doi.org/10.1080/02589001.2017.1307505
- Kebede, D., Emana, B., & Tesfay, G. (2023). Impact of land acquisition for large-scale agricultural investments on food security status of displaced households: The case of Ethiopia. Land Use Policy, 126, 106507.

https://doi.org/10.1016/j.landusepol.2022.106507

- Kebede, D., Tesfay, G., & Emana, B. (2021). Impact of land acquisition for large-scale agricultural investments on income and asset possession of displaced households in Ethiopia. *Heliyon*, 7(12), e08557. https://doi.org/10.1016/j.heliyon.2021.e08557
- Kinda, S. R. (2016). *Climatic Shocks and Food Security: The Role of Foreign Aid.* https://doi.org/10.2139/ssrn.2741725
- Knutsen, C. H., Kotsadam, A., Olsen, E. H., & Wig, T. (2017). Mining and Local Corruption in Africa. American Journal of Political Science, 61(2), 320–334. https://doi.org/10.1111/ajps.12268
- Land Matrix. (2024). Frequently asked questions (FAQs). Retrieved December 20, 2023, from https://landmatrix.org/faq/
- Maitra, P., & Tagat, A. (2024). Labor supply responses to rainfall shocks. *Review of Development Economics*. Advance online publication. https://doi.org/10.1111/rode.13079
- Meiyappan, P., & Jain, A. K. (2012). Three distinct global estimates of historical land-cover change and land-use conversions for over 200 years. *Frontiers of Earth Science*, 6(2), 122–139. https://doi.org/10.1007/s11707-012-0314-2

- Miller, R., Mudenda, L. D., & Sedai, A. K. (2024). Persistent Agricultural Shocks and Child Poverty. *The Journal of Development Studies*, 60(1), 30–45. https://doi.org/10.1080/00220388.2023.2253977
- Müller, M. F., Penny, G., Niles, M. T., Ricciardi, V., Chiarelli, D. D., Davis, K. F., Dell'Angelo, J., D'Odorico, P., Rosa, L., Rulli, M. C., & Mueller, N. D. (2021). Impact of transnational land acquisitions on local food security and dietary diversity. *Proceedings of the National Academy of Sciences of the United States of America*, *118*(4), e2020535118. https://doi.org/10.1073/pnas.2020535118
- Nübler, L., Austrian, K., Maluccio, J. A., & Pinchoff, J. (2021). Rainfall shocks, cognitive development and educational attainment among adolescents in a drought-prone region in Kenya. *Environment and Development Economics*, *26*(5-6), 466–487. https://doi.org/10.1017/s1355770x20000406
- Sändig, J. (2021). Contesting large-scale land acquisitions in the Global South. *World Development*, *146*, 105581. https://doi.org/10.1016/j.worlddev.2021.105581
- Schneider, Udo, Andreas Becker, Peter Finger, Anja Meyer-Christoffer, Bruno Rudolf. (2015). GPCC full data reanalysis version 7.0 (at 0.5, 1.0, 2.5): Monthly land-surface precipitation from rain-gauges built on GTS-based and historic data.
- Sekoai, P., & Yoro, K. (2016). Biofuel Development Initiatives in Sub-Saharan Africa: Opportunities and Challenges. *Climate*, 4(2), 33. https://doi.org/10.3390/cli4020033
- Sen, K., & Villa, K. (2022). Rainfall shocks and adolescent school-work transition: Evidence from rural South Africa.
 - https://ageconsearch.umn.edu/record/322383/files/24226.pdf
- Tamiru, D., Argaw, A., Gerbaba, M., Ayana, G., Nigussie, A., & Belachew, T. (2016).
 Household food insecurity and its association with school absenteeism among primary school adolescents in Jimma zone, Ethiopia. *BMC Public Health*, *16*(1), 802. https://doi.org/10.1186/s12889-016-3479-x
- Tollefsen, A. F., Strand, H., & Buhaug, H. (2012). Prio-GRID: A unified spatial data structure. Journal of Peace Research, 49(2), 363–374. https://doi.org/10.1177/0022343311431287
- Moreda, T. (2017). Listening to their silence? The political reaction of affected communities to large-scale land acquisitions: insights from Ethiopia. In *Global Land Grabbing and Political Reactions 'from Below'* (pp. 51–74). Routledge. https://doi.org/10.4324/9781315112565-3
- Vermeulen, S., & Cotula, L. (2010). Over the heads of local people: Consultation, consent, and recompense in large-scale land deals for biofuels projects in Africa. *The Journal* of Peasant Studies, 37(4), 899–916. https://doi.org/10.1080/03066150.2010.512463
- Wegenast, T., Richetta, C., Krauser, M., & Leibik, A. (2022). Grabbed trust? The impact of large-scale land acquisitions on social trust in Africa. *World Development*, 159, 106038. https://doi.org/10.1016/j.worlddev.2022.106038
- Yang, B., & He, J. (2021). Global Land Grabbing: A Critical Review of Case Studies across the World. *Land*, *10*(3), 324. https://doi.org/10.3390/land10030324

- Yengoh, G., & Armah, F. (2015). Effects of Large-Scale Acquisition on Food Insecurity in Sierra Leone. *Sustainability*, 7(7), 9505–9539. https://doi.org/10.3390/su7079505
- Yengoh, G. T., Steen, K., Armah, F. A., & Ness, B. (2016). Factors of vulnerability: How large-scale land acquisitions take advantage of local and national weaknesses in Sierra
 Leone. Land Use Policy, 50, 328–340.
 https://doi.org/10.1016/j.landusepol.2015.09.028

Appendix

Appendix A: Countries Included in the Analysis

Afrobarometer: Benin, Botswana, Burkina Faso, Cameroon, Cote d'Ivoire, Egypt, Gabon, Ghana, Guinea, Kenya, Lesotho, Liberia, Madagascar, Malawi, Mali, Morocco, Mozambique, Namibia, Nigeria, São Tomé and Príncipe, Senegal, Sierra Leone, South Africa, Sudan, Swaziland, Tanzania, Togo, Tunisia, Uganda, Zambia, Zimbabwe.

Land Matrix: Angola, Benin, Côte d'Ivoire, Cameroon, Rep. of Congo, Egypt, Ethiopia, Ghana, Guinea, Kenya, Liberia, Madagascar, Mali, Mozambique, Namibia, Nigeria, Sudan, South Sudan, Senegal, Sierra Leone, Swaziland, Togo, Tanzania, Uganda, Dem. Rep. Congo, Zambia, Gabon, Central African Republic, Chad, Zimbabwe, Malawi, São Tomé and Principe, Mauritania, Lesotho, Burkina Faso, Morocco, Botswana, South Africa, Tunisia

Appendix B: Regression Results

Note: By default, the *fixest* package does not report intercepts. Standard errors are clustered by region in all models, including those in the robustness checks and causal mechanism testing. In order to safe space, I do not provide the results of the F-tests in the appendix if any of these conditions apply: Both *active* and *inactive* are not statistically significant, or the difference between *active* and *inactive* is not significant. Exceptions where these conditions do not apply are addressed in the main text and appendix, along with the corresponding F-test results. I do not provide F-tests for the interaction models, as the focus there is on the interaction term between *active* and *gender*. I would provide them for the gender-differentiated data sets. However, in all cases, either one of the coefficients is not significant or the F-test result is not significant.

	Education
Active	0.2010*
	(0.0870)
Inactive	0.5576*
	(0.2176)
Gender	0.5435***
	(0.0173)
Religion	-0.4573***
	(0.0318)
Urban	0.0576***
	(0.0160)
Drought	-1.3397*
	(0.5352)
Rain	0.0003
	(0.0003)
Agricultural Land	-0.0002
	(0.0015)
Exposure Time	-0.0277*
	(0.0126)
Age (18 - 25)	0.8058***
	(0.0362)
Age (26 - 35)	0.6869***
	(0.0331)
Age (36 - 45)	0.3531***
	(0.0333)
Age (56 - 65)	-0.3982***
	(0.0419)
Age (66 +)	-0.8981***
	(0.0503)
Num.Obs.	43102
R2	0.315
R2 Adj.	0.308
AIC	172897.6
BIC	176452.8
RMSE	1.78
Std.Errors	Heteroskedasticity-robust
FE: region	Yes
• + p < 0.1, * p	< 0.05, ** p < 0.01, *** p < 0.001

Table B.1: Regression Results

	Education
Active	0.2395*
	(0.1002)
Inactive	0.5578*
	(0.2176)
Gender	0.5450***
	(0.0175)
Religion	-0.4573***
	(0.0318)
Urban	0.0576***
	(0.0160)
Drought	-1.3381*
	(0.5352)
Rain	0.0003
	(0.0003)
Agricultural Land	-0.0002
-	(0.0015)
Exposure Time	-0.0278*
-	(0.0125)
Age (18 - 25)	0.8059***
	(0.0362)
Age (26 - 35)	0.6869***
	(0.0331)
Age (36 – 45)	0.3531***
	(0.0333)
Age 56 - 65)	-0.3983***
0 ,	(0.0419)
Age (66 +)	-0.8983***
	(0.0503)
Active × Gender	-0.0806
	(0.1104)
Num.Obs.	43102
R2	0.315
R2 Adj.	0.308
AIC	172899.2
BIC	176463.1
RMSE	1.78
Std.Errors	Heteroskedasticitv-robust

Table B.2. : Regression Results (Interaction)

F – Test: F = 2.71, p < 0.1

	(Male)	(Female) Education
Activo	0.2321+	0 1 4 5 5
Active	(0.1235)	(0.1230)
Inactivo	0.1233)	0.6609+
macuve	(0.3714)	(0.2497)
Deligion	0.4720***	0.3467
Religion	-0.4/29	-0.4705
TT]	(0.0453)	(0.0444)
Urban	0.0480*	0.0666**
	(0.0232)	(0.0219)
Drought	-1.4541+	-0.9296
	(0.7596)	(0.7553)
Rain	0.0000	0.0004
	(0.0004)	(0.0004)
Agricultural Land	-0.0005	0.0005
	(0.0021)	(0.0021)
Exposure Time	-0.0267+	-0.0279
	(0.0157)	(0.0195)
Age (18 - 25)	0.6888***	0.9571***
	(0.0526)	(0.0500)
Age (26 - 35)	0.5969***	0.7922***
	(0.0481)	(0.0455)
Age (36 - 45)	0.3179***	0.4044***
	(0.0479)	(0.0464)
Age (56 - 65)	-0.3273***	-0.5329***
	(0.0578)	(0.0605)
Age (66 +)	-0.8586***	-1.0558***
	(0.0673)	(0.0763)
Num.Obs.	21352	21750
R2	0.290	0.353
R2 Adj.	0.276	0.341
AIC	86408.3	86279.5
BIC	89667.5	89546.3
RMSE	1.80	1.73
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes

Table B.3. : Regression Results (Gender Differentiated)

Appendix C: Robustness Checks

Note: I also report the models with a 10 km buffer. However, my confidence in the 10 km buffer is limited, as only 0.39% of respondents are coded as *active* and 0.08% as *inactive* at this buffer size.

Primary Education and *Secondary Education* are binary variables. *Primary Education* is coded as 1 if the respondent indicates having some primary schooling or having completed primary schooling, and 0 otherwise. *Secondary Education* is coded as 1 if the respondent reports attending intermediate school, some secondary school/high school, having completed secondary school/high school, or having post-secondary qualifications other than a university degree, such as a diploma from a polytechnic or college, and 0 otherwise.

	(10 km)	(25 km)	(50 km)
	Education	Education	Education
Active	0.4122+	0.2010*	0.0758+
	(0.2142)	(0.0870)	(0.0432)
Inactive	0.5184	0.5576*	0.4438***
	(0.3287)	(0.2176)	(0.1198)
Gender	0.5435***	0.5435***	0.5433***
	(0.0173)	(0.0173)	(0.0173)
Religion	-0.4520***	-0.4573***	-0.4549***
	(0.0317)	(0.0318)	(0.0317)
Urban	0.0585***	0.0576***	0.0584***
	(0.0157)	(0.0160)	(0.0157)
Drought	-1.3518*	-1.3397*	-1.2929*
	(0.5338)	(0.5352)	(0.5343)
Rain	0.0003	0.0003	0.0003
	(0.0003)	(0.0003)	(0.0003)
Agri. Land	-0.0001	-0.0002	-0.0003
	(0.0015)	(0.0015)	(0.0015)
Exp. Time	-0.0123	-0.0277*	-0.0082
	(0.0294)	(0.0126)	(0.0075)
Age (18 - 25)	0.8103***	0.8058***	0.7976***
	(0.0359)	(0.0362)	(0.0364)
Age (26 - 35)	0.6882***	0.6869***	0.6834***
	(0.0330)	(0.0331)	(0.0331)
Age (36 - 45)	0.3541***	0.3531***	0.3523***
	(0.0333)	(0.0333)	(0.0333)
Age (56 – 65)	-0.3974***	-0.3982***	-0.3975***
	(0.0419)	(0.0419)	(0.0419)
Age (66 +)	-0.8975***	-0.8981***	-0.8983***
	(0.0503)	(0.0503)	(0.0503)
Num.Obs.	43317	43102	43232
R2	0.314	0.315	0.314
R2 Adj.	0.307	0.308	0.308
AIC	173718.4	172897.6	173400.1
BIC	177275.7	176452.8	176956.6
RMSE	1.78	1.78	1.78
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table C.1 : Comparative Regression Results

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

F – Test (25 km): F = 2.71, p < 0.1

F – Test (50 km): F = 9.04, p < 0.01

	(10 km)	(25 km)	(50 km)
	Education	Education	Education
Active	0.5198*	0.2395*	0.0615
	(0.2341)	(0.1002)	(0.0528)
Inactive	0.5123	0.5578*	0.4439***
	(0.3294)	(0.2176)	(0.1198)
Gender	0.5444***	0.5450***	0.5413***
	(0.0173)	(0.0175)	(0.0180)
Religion	-0.4520***	-0.4573***	-0.4549***
	(0.0317)	(0.0318)	(0.0317)
Urban	0.0585***	0.0576***	0.0583***
	(0.0157)	(0.0160)	(0.0157)
Drought	-1.3523*	-1.3381*	-1.2954*
	(0.5339)	(0.5352)	(0.5344)
Rain	0.0003	0.0003	0.0003
	(0.0003)	(0.0003)	(0.0003)
Agri. Land	-0.0001	-0.0002	-0.0003
	(0.0015)	(0.0015)	(0.0015)
Exp. Time	-0.0108	-0.0278*	-0.0082
	(0.0297)	(0.0125)	(0.0075)
Age (18 - 25)	0.8102***	0.8059***	0.7976***
	(0.0359)	(0.0362)	(0.0364)
Age (26 -35)	0.6883***	0.6869***	0.6834***
	(0.0330)	(0.0331)	(0.0331)
Age (36 - 45)	0.3541***	0.3531***	0.3524***
	(0.0333)	(0.0333)	(0.0333)
Age (56 - 65)	-0.3975***	-0.3983***	-0.3973***
	(0.0419)	(0.0419)	(0.0419)
Age (66 +)	-0.8977***	-0.8983***	-0.8980***
	(0.0503)	(0.0503)	(0.0503)
Active × Gender	-0.2530	-0.0806	0.0305
	(0.2631)	(0.1104)	(0.0629)
Num.Obs.	43317	43102	43232
R2	0.314	0.315	0.314
R2 Adj.	0.307	0.308	0.308
AIC	173719.6	172899.2	173401.9
BIC	177285.5	176463.1	176967.1
RMSE	1.78	1.78	1.78
Std.Errors	Heteroskedasticity- robust	Heteroskedasticity- robust	Heteroskedasticity- robust
FE: region	Yes	Yes	Yes

Table C.2 : Comparative Regression Results (Interaction)

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

	(M: 10 km) Education	(M:25 km) Education	(M: 50 km) Education	(F: 10 km) Education	(F: 25 km) Education	(F: 50 km) Education
Active	0.4633	0.2321+	0.1240*	0.3437	0.1455	0.0273
	(0.3447)	(0.1235)	(0.0613)	(0.2722)	(0.1239)	(0.0606)
Inactive	0.6516	0.5371*	0.2247	0.5338	0.6609+	0.6201***
	(0.4507)	(0.2714)	(0.1605)	(0.4793)	(0.3487)	(0.1747)
Religion	-0.4701***	-0.4729***	-0.4733***	-0.4628***	-0.4705***	-0.4647***
	(0.0452)	(0.0453)	(0.0452)	(0.0444)	(0.0444)	(0.0445)
Urban	0.0497*	0.0480*	0.0481*	0.0674**	0.0666**	0.0683**
	(0.0228)	(0.0232)	(0.0229)	(0.0214)	(0.0219)	(0.0214)
Drought	-1.4469+	-1.4541+	-1.4228+	-0.9636	-0.9296	-0.8624
	(0.7577)	(0.7596)	(0.7593)	(0.7532)	(0.7553)	(0.7533)
Rain	0.0000	0.0000	-0.0001	0.0003	0.0004	0.0003
	(0.0004)	(0.0004)	(0.0004)	(0.0004)	(0.0004)	(0.0004)
Agri. Land	-0.0006	-0.0005	-0.0007	0.0007	0.0005	0.0004
	(0.0021)	(0.0021)	(0.0021)	(0.0021)	(0.0021)	(0.0021)
Exp. Time	-0.0367	-0.0267+	-0.0091	0.0137	-0.0279	-0.0072
	(0.0419)	(0.0157)	(0.0102)	(0.0418)	(0.0195)	(0.0110)
Age (18- 26)	0.6917***	0.6888***	0.6780***	0.9608***	0.9571***	0.9521***
	(0.0522)	(0.0526)	(0.0529)	(0.0496)	(0.0500)	(0.0503)
Age (26 - 35)	0.5991***	0.5969***	0.5941***	0.7913***	0.7922***	0.7873***
	(0.0480)	(0.0481)	(0.0481)	(0.0454)	(0.0455)	(0.0455)
Age (36 - 45)	0.3193***	0.3179***	0.3180***	0.4047***	0.4044***	0.4028***
	(0.0478)	(0.0479)	(0.0478)	(0.0464)	(0.0464)	(0.0464)
Age (46 -55)	-0.3268***	-0.3273***	-0.3263***	-0.5320***	-0.5329***	-0.5329***
	(0.0578)	(0.0578)	(0.0578)	(0.0605)	(0.0605)	(0.0605)
Age (66+)	-0.8590***	-0.8586***	-0.8577***	-1.0540***	-1.0558***	-1.0570***
	(0.0672)	(0.0673)	(0.0673)	(0.0762)	(0.0763)	(0.0763)
Num.Ob s.	21448	21352	21410	21869	21750	21822
R2	0.290	0.290	0.290	0.352	0.353	0.353
R2 Adj.	0.276	0.276	0.276	0.340	0.341	0.341
AIC	86767.0	86408.3	86635.0	86741.3	86279.5	86549.4
BIC	90028.1	89667.5	89895.4	90010.4	89546.3	89817.6
RMSE	1.79	1.80	1.80	1.73	1.73	1.73
Std.Erro rs	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust
FE: region	Yes	Yes	Yes	Yes	Yes	Yes

Table C.3 : Comparative Regression Results (Gender Differentiated)

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. M = Male. F = Female.

	(10 km)	(25 km)	(50 km)
	Primary Education	Primary Education	Primary Education
Active	-0.5071	-0.2166	-0.1394*
	(0.3144)	(0.1336)	(0.0613)
Inactive	-0.7201	-0.6438	-0.1636
	(0.5853)	(0.4091)	(0.1615)
Gender	-0.2411***	-0.2383***	-0.2392***
	(0.0231)	(0.0231)	(0.0231)
Religion	0.0557	0.0556	0.0549
	(0.0412)	(0.0413)	(0.0413)
Urban	-0.0465+	-0.0432+	-0.0464+
	(0.0237)	(0.0240)	(0.0238)
Drought	0.9941	1.0370	1.0317
	(0.7191)	(0.7211)	(0.7193)
Rain	-0.0025***	-0.0024***	-0.0025***
	(0.0004)	(0.0004)	(0.0004)
Agri. Land	0.0016	0.0016	0.0017
	(0.0019)	(0.0019)	(0.0019)
Exp. Time	0.0290	0.0219	0.0007
	(0.0411)	(0.0189)	(0.0111)
Age (18 - 25)	-0.5358***	-0.5352***	-0.5149***
	(0.0482)	(0.0485)	(0.0489)
Age (26 - 35)	-0.3207***	-0.3233***	-0.3140***
	(0.0419)	(0.0420)	(0.0420)
Age (36 - 45)	-0.2149***	-0.2125***	-0.2136***
	(0.0421)	(0.0421)	(0.0421)
Age (56 - 55)	0.1011+	0.1007+	0.1001+
	(0.0527)	(0.0527)	(0.0527)
Age (66 +)	0.0303	0.0301	0.0288
	(0.0648)	(0.0648)	(0.0648)
Num.Obs.	43195	42980	43110
R2	0.130	0.130	0.130
R2 Adj.	0.115	0.115	0.115
AIC	46834.6	46616.5	46753.8
BIC	50338.7	50118.6	50257.1
RMSE	0.42	0.42	0.42
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table C.4 : Logit Regression Results Primary Education

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

	(10 km)	(25 km)	(50 km)
	Primary Education	Primary Education	Primary Education
Active	-0.4635	-0.1360	-0.0270
	(0.3493)	(0.1501)	(0.0735)
Inactive	-0.7216	-0.6433	-0.1632
	(0.5858)	(0.4093)	(0.1616)
Gender	-0.2408***	-0.2350***	-0.2232***
	(0.0231)	(0.0234)	(0.0239)
Religion	0.0556	0.0557	0.0550
	(0.0412)	(0.0413)	(0.0413)
Urban	-0.0465+	-0.0431+	-0.0459+
	(0.0237)	(0.0240)	(0.0238)
Drought	0.9937	1.0403	1.0506
	(0.7191)	(0.7212)	(0.7194)
Rain	-0.0025***	-0.0024***	-0.0024***
	(0.0004)	(0.0004)	(0.0004)
Agri. Land	0.0016	0.0016	0.0017
	(0.0019)	(0.0019)	(0.0019)
Exp. Time	0.0293	0.0216	0.0005
	(0.0413)	(0.0190)	(0.0112)
Age (18 - 26)	-0.5359***	-0.5351***	-0.5148***
	(0.0482)	(0.0485)	(0.0489)
Age (26 - 35)	-0.3207***	-0.3232***	-0.3140***
	(0.0419)	(0.0420)	(0.0420)
Age (36 - 45)	-0.2149***	-0.2125***	-0.2138***
	(0.0421)	(0.0421)	(0.0421)
Age (56 - 65)	0.1011+	0.1005+	0.0993+
	(0.0527)	(0.0527)	(0.0527)
Age (66+)	0.0302	0.0296	0.0269
	(0.0648)	(0.0648)	(0.0648)
Active × Gender	-0.1101	-0.1810	-0.2599**
	(0.4133)	(0.1698)	(0.0962)
Num.Obs.	43195	42980	43110
R2	0.130	0.130	0.130
R2 Adj.	0.115	0.115	0.115
AIC	46836.6	46617.4	46748.5
BIC	50349.3	50128.2	50260.4
RMSE	0.42	0.42	0.42
Std.Errors	Heteroskedasticity- robust	Heteroskedasticity- robust	Heteroskedasticity- robust
FE: region	Yes	Yes	Yes

Table C. 5: Logit Regression Results Primary Education (Interaction)

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

	(M:10 km)	(M: 25 km)	(M:50 km)	(F: 10 km)	(F: 25 km)	(F: 50 km)
	Primary Education	Primary Education	Primary Education	Primary Education	Primary Education	Primary Education
Active	-0.1505	-0.0029	-0.2245*	-0.6583	-0.3494+	-0.0824
	(0.4983)	(0.2009)	(0.0959)	(0.4077)	(0.1842)	(0.0820)
Inactive	-10.6646***	-1.7490+	-0.4319	-0.2900	-0.1705	-0.0290
	(0.3042)	(1.0379)	(0.2750)	(0.6908)	(0.5249)	(0.2100)
Religion	0.1091+	0.1044+	0.1094+	0.0037	0.0093	0.0032
	(0.0600)	(0.0601)	(0.0602)	(0.0584)	(0.0587)	(0.0584)
Urban	-0.0535	-0.0513	-0.0516	-0.0452	-0.0408	-0.0459
	(0.0405)	(0.0407)	(0.0406)	(0.0301)	(0.0306)	(0.0302)
Drought	2.4821*	2.6264*	2.5002*	-0.2460	-0.3039	-0.1814
	(1.0443)	(1.0445)	(1.0447)	(1.0222)	(1.0275)	(1.0229)
Rain	-0.0020***	-0.0020***	-0.0020***	-0.0028***	-0.0028***	-0.0028***
	(0.0005)	(0.0005)	(0.0005)	(0.0006)	(0.0006)	(0.0006)
Agri. Land	0.0004	0.0002	0.0005	0.0027	0.0029	0.0028
	(0.0029)	(0.0029)	(0.0029)	(0.0026)	(0.0026)	(0.0026)
Exp. Time	-0.0296	-0.0307	-0.0037	0.0544	0.0579*	0.0072
	(0.0696)	(0.0311)	(0.0175)	(0.0526)	(0.0250)	(0.0147)
Age (18 - 25)	-0.6886***	-0.6836***	-0.6497***	-0.4157***	-0.4202***	-0.4095***
	(0.0715)	(0.0721)	(0.0725)	(0.0678)	(0.0683)	(0.0688)
Age (26 - 35)	-0.3538***	-0.3532***	-0.3422***	-0.2866***	-0.2928***	-0.2856***
	(0.0611)	(0.0613)	(0.0612)	(0.0597)	(0.0598)	(0.0598)
Age (36 -45)	-0.2413***	-0.2409***	-0.2405***	-0.1916**	-0.1880**	-0.1914**
	(0.0600)	(0.0601)	(0.0600)	(0.0607)	(0.0607)	(0.0607)
Age (56 -65)	0.0880	0.0864	0.0858	0.1030	0.1044	0.1032
	(0.0722)	(0.0722)	(0.0721)	(0.0798)	(0.0798)	(0.0798)
Age (66 +)	0.0063	0.0034	0.0033	0.0280	0.0308	0.0265
	(0.0862)	(0.0863)	(0.0862)	(0.1035)	(0.1035)	(0.1035)
Num.Ob s.	21300	21204	21262	21688	21569	21641
R2	0.144	0.145	0.144	0.129	0.129	0.128
R2 Adj.	0.113	0.113	0.113	0.100	0.100	0.099
AIC	22529.1	22440.4	22498.6	24557.7	24422.1	24509.2
BIC	25691.8	25601.3	25660.6	27711.6	27573.8	27662.2
RMSE	0.41	0.41	0.41	0.43	0.43	0.43
Std.Erro rs	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust
FE: region	Yes	Yes	Yes	Yes	Yes	Yes

Table C.6 : Logit Regression Results Primary Education (Gender Differentiated)

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. M = Male. F = Female.

	(10 km)	(25 km)	(50 km)
	Secondary Education	Secondary Education	Secondary Education
Active	0.0288	-0.0021	0.2362***
	(0.2764)	(0.1191)	(0.0558)
Inactive	0.1569	0.6158+	0.0485
	(0.4782)	(0.3379)	(0.1403)
Gender	0.4024***	0.4007***	0.4015***
	(0.0220)	(0.0221)	(0.0220)
Religion	-0.3916***	-0.3951***	-0.3950***
	(0.0401)	(0.0402)	(0.0402)
Urban	-0.0112	-0.0128	-0.0113
	(0.0169)	(0.0171)	(0.0170)
Drought	-2.6833***	-2.7446***	-2.7374***
	(0.6848)	(0.6867)	(0.6866)
Rain	0.0017***	0.0017***	0.0016***
	(0.0003)	(0.0003)	(0.0003)
Agri. Land	0.0050**	0.0051**	0.0049**
	(0.0017)	(0.0017)	(0.0017)
Exp. Time	0.0129	0.0028	0.0053
	(0.0385)	(0.0170)	(0.0097)
Age (18 -25)	0.9906***	0.9840***	0.9478***
	(0.0444)	(0.0447)	(0.0451)
Age (26 - 35)	0.5606***	0.5592***	0.5468***
	(0.0405)	(0.0406)	(0.0406)
Age (36 - 45)	0.4420***	0.4412***	0.4389***
	(0.0417)	(0.0417)	(0.0417)
Age (56 - 65)	-0.3567***	-0.3565***	-0.3545***
	(0.0539)	(0.0540)	(0.0539)
Age (66+)	-0.9064***	-0.9058***	-0.9033***
	(0.0715)	(0.0715)	(0.0716)
Num.Obs.	43218	43003	43133
R2	0.162	0.162	0.162
R2 Adj.	0.148	0.148	0.149
AIC	50497.3	50220.7	50351.7
BIC	54018.9	53740.3	53872.5
RMSE	0.44	0.44	0.44
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table C.7 : Logit Regression Results Secondary Education

• +p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

Secondary EducationSecondary EducationSecondary EducationActive0.2067-0.03280.1573*(0.3108)(0.1357)(0.0672)Inactive0.15130.6156+0.0482(0.4772)(0.3379)(0.1403)Gender0.4041***0.3993***0.3894***(0.0220)(0.0223)(0.0228)Religion-0.3915***-0.3952***-0.3950***(0.0401)(0.0402)(0.0402)Urban-0.0111-0.0128-0.0115(0.0648)(0.0171)(0.0170)Drought-2.6846***-2.7455***-2.7477***(0.6848)(0.6867)(0.6867)Rain0.0017***0.0017***0.0016***(0.0003)(0.0003)(0.0003)(0.0003)Agri. Land0.0050**0.0051**0.0049**(0.080)(0.0169)(0.0098)Age (18 - 25)0.9905***(0.0405)(0.0444)(0.0447)(0.0451)Age (26 - 35)0.5607***0.5591***0.5465***(0.0405)(0.0406)(0.0406)Age (36 - 45)(0.417)(0.0417)(0.0417)(0.0417)Age (56 - 5)-0.3569***-0.3564***-0.3538**(0.0539)(0.0540)(0.0539)(0.0540)(0.0715)(0.0715)(0.0716)Active \times (0.0715)(0.0715)(0.0716)Active \times (0.0715)(0.0715)(0.0716)Age (66+)-0.9067***-0.9056***(0.0715)(0.07		(10 km)	(25 km)	(50 km)
Active 0.2067 -0.0328 0.1573* Inactive 0.1513 0.6156+ 0.0482 (0.4772) (0.3379) (0.1403) Gender 0.4041*** 0.3993*** 0.3894*** (0.0220) (0.0223) (0.028) Religion -0.3915*** -0.3952*** -0.3950*** (0.0401) (0.0402) (0.0402) Urban -0.0111 -0.0128 -0.0115 (0.0401) (0.0171) (0.0170) Drought -2.6846*** -2.7455*** -2.7477*** (0.6848) (0.6867) (0.6867) Rain 0.0017*** 0.0016*** 0.0017 Quo03) (0.003) (0.0003) Qu003 Agri. Land 0.005** 0.0051** 0.0049** (0.0441) (0.017) (0.017) Qu017) Exp. Time 0.141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Qage (36 - 45) 0.4420*** 0.44047) (0.0417) A		Secondary Education	Secondary Education	Secondary Education
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Active	0.2067	-0.0328	0.1573*
Inactive 0.1513 0.6156+ 0.0482 (0.4772) (0.3379) (0.1403) Gender 0.4041^{***} 0.3993^{***} 0.3894^{***} (0.0220) (0.0223) (0.0228) Religion -0.3915^{***} -0.3952^{***} -0.3950^{***} (0.0401) (0.0402) (0.0402) Urban -0.0111 -0.0128 -0.115 (0.069) (0.0171) (0.0170) Drought -2.6846^{***} -2.7455^{***} -2.7477^{***} (0.6848) (0.6867) (0.6867) Rain 0.0017^{***} 0.0016^{***} 0.0016^{***} (0.003) (0.0003) (0.0003) 0.0003 Agri. Land 0.005^{***} 0.0051^{**} 0.0049^{**} (0.0017) (0.0017) (0.0017) 0.0054 (0.380) (0.0169) (0.098) Age (18 - 25) 0.9905^{***} 0.9840^{***} 0.9478^{***} (0.417) (0.444) $(0.041$		(0.3108)	(0.1357)	(0.0672)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Inactive	0.1513	0.6156+	0.0482
Gender 0.4041*** 0.3993*** 0.3894*** (0.0220) (0.0223) (0.0228) Religion -0.3915*** -0.3952*** -0.3950*** (0.0401) (0.0402) (0.0402) Urban -0.0111 -0.0128 -0.0115 (0.0169) (0.0171) (0.0170) Drought -2.6846*** -2.7455*** -2.7477*** (0.6848) (0.6867) (0.6867) Rain 0.0017*** 0.0016*** 0.0016*** (0.0003) (0.0003) (0.0003) (0.0003) Agri. Land 0.0050** 0.0029 0.0054 (0.0017) (0.0017) (0.0017) (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Age (26 - 35) 0.5607*** 0.5401*** 0.5465*** (0.0417) (0.0417) (0.0417) (0.0417) Age (26 - 55) 0.3569*** 0.3564*** 0.3538** (0.0539) (0.0540) <		(0.4772)	(0.3379)	(0.1403)
(0.0220) (0.0223) (0.0228) Religion -0.3915*** -0.3952*** -0.3950*** (0.0401) (0.0402) (0.0402) Urban -0.0111 -0.0128 -0.0115 (0.0169) (0.0171) (0.0170) Drought -2.6846*** -2.7455*** -2.7477*** (0.66848) (0.6867) (0.6867) Rain 0.0017*** 0.0017*** 0.0016*** (0.0003) (0.0003) (0.0003) Agri. Land 0.0050** 0.0017 (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0044) (0.0447) (0.0451) Age (18 - 25) 0.9905*** 0.9840*** 0.9478*** (0.0444) (0.0447) (0.0451) Age (26 - 35) 0.5607*** 0.5591*** 0.5465*** (0.0405) (0.0406) (0.0417) Age (26 - 55) 0.3569*** 0.3564*** 0.3538** (0.0417) (0.0417) (0.0417) Age (56 - 65)	Gender	0.4041***	0.3993***	0.3894***
Religion -0.3915*** -0.3952*** -0.3950*** (0.0401) (0.0402) (0.0402) Urban -0.0111 -0.0128 -0.0115 (0.0169) (0.0171) (0.0170) Drought -2.6846*** -2.7455*** -2.7477*** (0.6848) (0.6867) (0.6867) Rain 0.0017*** 0.0017*** 0.0016*** (0.0003) (0.0003) (0.0003) (0.0003) Agri. Land 0.0050** 0.0017) (0.0017) (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Age (18 - 25) 0.9905*** 0.9840*** 0.9478*** (0.0444) (0.0447) (0.0451) Age (26 - 35) 0.5607*** 0.5591*** 0.5465*** (0.0405) (0.0446) (0.0406) (0.0406) Age (26 - 55) 0.5609*** 0.3564*** 0.3538*** (0.0539) (0.0539) (0.0540) (0.0539) Age (66+) <td></td> <td>(0.0220)</td> <td>(0.0223)</td> <td>(0.0228)</td>		(0.0220)	(0.0223)	(0.0228)
(0.0401) (0.0402) (0.0402) Urban -0.0111 -0.0128 -0.0115 (0.0169) (0.0171) (0.0170) Drought -2.6846*** -2.7455*** -2.7477*** (0.6848) (0.6867) (0.6867) Rain 0.0017*** 0.0016*** (0.0003) (0.0003) (0.0003) Agri. Land 0.0050** 0.0051** 0.0049** (0.0017) (0.0017) (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Age (18 - 25) 0.9905*** 0.9840*** 0.9478*** (0.0444) (0.0447) (0.0451) Age (26 - 35) 0.5607*** 0.5591*** 0.5465*** (0.0405) 0.591*** 0.4390*** 0.4412*** Age (36 - 45) 0.4420*** 0.4412*** 0.4390*** (0.0539) (0.0540) (0.0539) 0.0546*** (0.0715) (0.0716) (0.0716) Age (66+)	Religion	-0.3915***	-0.3952***	-0.3950***
Urban -0.0111 -0.0128 -0.0115 0.0169) (0.0171) (0.0170) Drought -2.6846*** -2.7455*** -2.7477*** 0.6848) (0.6867) (0.6867) Rain 0.0017*** 0.0017*** 0.0016*** 0.0003) (0.0003) (0.0003) Agri. Land 0.0050** 0.0051** 0.0049** (0.0017) (0.0017) (0.0017) (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Age (18 - 25) 0.9905** 0.9840*** 0.9478** (0.0444) (0.0447) (0.0451) Age (26 - 35) 0.5607*** 0.5591*** 0.5465*** (0.0405) (0.0406) (0.0406) 0.412** Age (36 - 45) 0.4420*** 0.4412*** 0.4390*** (0.0539) (0.0540) (0.0539) 0.956*** (0.0539) (0.0540) (0.0539) 0.9017*** (0.0715) (0.0715)		(0.0401)	(0.0402)	(0.0402)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Urban	-0.0111	-0.0128	-0.0115
$\begin{array}{llllllllllllllllllllllllllllllllllll$		(0.0169)	(0.0171)	(0.0170)
Rain (0.6848) (0.6867) (0.6867) Rain 0.0017*** 0.0017*** 0.0016*** (0.003) (0.0003) (0.0003) Agri. Land 0.0050** 0.0051** 0.0049** (0.0017) (0.0017) (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Age (18 - 25) 0.9905*** 0.9840*** 0.9478*** (0.0444) (0.0447) (0.0451) Age (26 - 35) 0.5607*** 0.5591*** 0.5465*** (0.0405) (0.0406) (0.0406) (0.0406) Age (26 - 35) 0.4420*** 0.4412*** 0.4390*** (0.0417) (0.0417) (0.0417) Age (56 - 65) -0.3569*** -0.3564*** -0.3538*** (0.0539) (0.0540) (0.0539) Age (66+) -0.9067*** -0.9056*** -0.9017*** (0.0715) (0.0715) (0.0716) (0.0716)	Drought	-2.6846***	-2.7455***	-2.7477***
Rain 0.0017^{***} 0.0017^{***} 0.0016^{***} (0.003) (0.003) (0.003) Agri. Land 0.0050^{**} 0.0051^{**} 0.0049^{**} (0.0017) (0.0017) (0.0017) (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Age $(18 - 25)$ 0.9905^{***} 0.9840^{***} 0.9478^{***} (0.0444) (0.0447) (0.0451) Age $(26 - 35)$ 0.5607^{***} 0.5591^{***} 0.5465^{***} (0.0405) (0.0406) (0.0406) Age $(36 - 45)$ 0.4420^{***} 0.4412^{***} 0.4390^{***} (0.0417) (0.0417) (0.0417) (0.0417) Age $(56 - 65)$ -0.3569^{***} -0.3564^{***} -0.3538^{***} (0.0539) (0.0540) (0.0539) Age $(66+)$ -0.9067^{***} -0.9056^{***} -0.9017^{***} (0.0715) (0.0715) (0.0716) Active \times -0.4148 0.0662 0.0979		(0.6848)	(0.6867)	(0.6867)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Rain	0.0017***	0.0017***	0.0016***
Agri. Land 0.0050** 0.0051** 0.0049** (0.0017) (0.0017) (0.0017) Exp. Time 0.0141 0.0029 0.0054 (0.0380) (0.0169) (0.0098) Age (18 - 25) 0.9905*** 0.9840*** 0.9478*** (0.0444) (0.0447) (0.0451) Age (26 - 35) 0.5607*** 0.5591*** 0.5465*** (0.0405) (0.0406) (0.0406) (0.0406) Age (36 - 45) 0.4420*** 0.4412*** 0.4390*** (0.0417) (0.0417) (0.0417) (0.0417) Age (56 - 65) -0.3569*** -0.3564*** -0.3538*** (0.0539) (0.0540) (0.0539) Age (66+) -0.9067*** -0.9056*** -0.9017*** (0.0715) (0.0715) (0.0716) Active × -0.4148 0.0662 0.1756*		(0.0003)	(0.0003)	(0.0003)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Agri. Land	0.0050**	0.0051**	0.0049**
Exp. Time0.01410.00290.0054 (0.0380) (0.0169) (0.0098) Age $(18 - 25)$ 0.9905^{***} 0.9840^{***} 0.9478^{***} (0.0444) (0.0447) (0.0451) Age $(26 - 35)$ 0.5607^{***} 0.5591^{***} 0.5465^{***} (0.0405) (0.0406) (0.0406) Age $(36 - 45)$ 0.4420^{***} 0.4412^{***} 0.4390^{***} (0.0417) (0.0417) (0.0417) Age $(56 - 65)$ -0.3569^{***} -0.3564^{***} -0.3538^{***} (0.0539) (0.0540) (0.0539) Age $(66+)$ -0.9067^{***} -0.9056^{***} -0.9017^{***} (0.0715) (0.0715) (0.0716) Active \times Gender -0.4148 0.0662 0.1756^{*}		(0.0017)	(0.0017)	(0.0017)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Exp. Time	0.0141	0.0029	0.0054
Age $(18 - 25)$ 0.9905***0.9840***0.9478*** (0.0444) (0.0447) (0.0451) Age $(26 - 35)$ 0.5607^{***} 0.5591^{***} 0.5465^{***} (0.0405) (0.0406) (0.0406) Age $(36 - 45)$ 0.4420^{***} 0.4412^{***} 0.4390^{***} (0.0417) (0.0417) (0.0417) Age $(56 - 65)$ -0.3569^{***} -0.3564^{***} -0.3538^{***} (0.0539) (0.0540) (0.0539) Age $(66+)$ -0.9067^{***} -0.9056^{***} -0.9017^{***} (0.0715) (0.0715) (0.0716) Active \times Gender -0.4148 0.0662 0.1756^{*}		(0.0380)	(0.0169)	(0.0098)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Age (18 – 25)	0.9905***	0.9840***	0.9478***
Age $(26 - 35)$ 0.5607***0.5591***0.5465*** (0.0405) (0.0406) (0.0406) Age $(36 - 45)$ 0.4420***0.4412***0.4390*** (0.0417) (0.0417) (0.0417) Age $(56 - 65)$ -0.3569^{***} -0.3564^{***} -0.3538^{***} (0.0539) (0.0540) (0.0539) Age $(66+)$ -0.9067^{***} -0.9056^{***} -0.9017^{***} (0.0715) (0.0715) (0.0716) Active × Gender -0.4148 0.0662 0.1756^{*}		(0.0444)	(0.0447)	(0.0451)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Age (26 - 35)	0.5607***	0.5591***	0.5465***
Age $(36 - 45)$ 0.4420^{***} 0.4412^{***} 0.4390^{***} (0.0417) (0.0417) (0.0417) Age $(56 - 65)$ -0.3569^{***} -0.3564^{***} -0.3538^{***} (0.0539) (0.0540) (0.0539) Age $(66+)$ -0.9067^{***} -0.9056^{***} -0.9017^{***} (0.0715) (0.0715) (0.0716) Active × Gender -0.4148 0.0662 0.1756^{*}		(0.0405)	(0.0406)	(0.0406)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Age (36 - 45)	0.4420***	0.4412***	0.4390***
Age (56 - 65) -0.3569^{***} -0.3564^{***} -0.3538^{***} (0.0539)(0.0540)(0.0539)Age (66+) -0.9067^{***} -0.9056^{***} -0.9017^{***} (0.0715)(0.0715)(0.0716)Active × Gender -0.4148 0.0662 0.1756^{*}		(0.0417)	(0.0417)	(0.0417)
(0.0539) (0.0540) (0.0539) Age (66+) -0.9067*** -0.9056*** -0.9017*** (0.0715) (0.0715) (0.0716) Active × Gender -0.4148 0.0662 0.1756*	Age (56 - 65)	-0.3569***	-0.3564***	-0.3538***
Age (66+) -0.9067*** -0.9056*** -0.9017*** (0.0715) (0.0715) (0.0716) Active × -0.4148 0.0662 0.1756* Gender (0.2517) (0.1544) (0.0878)		(0.0539)	(0.0540)	(0.0539)
(0.0715) (0.0715) (0.0716) Active × -0.4148 0.0662 0.1756* Gender (0.2517) (0.1544) (0.0878)	Age (66+)	-0.9067***	-0.9056***	-0.9017***
Active × -0.4148 0.0662 0.1756* Gender (0.2517) (0.1544) (0.0878)		(0.0715)	(0.0715)	(0.0716)
(0.2517) (0.1544) (0.0979)	Active × Gender	-0.4148	0.0662	0.1756*
		(0.3517)	(0.1544)	(0.0878)
Num.Obs. 43218 43003 43133	Num.Obs.	43218	43003	43133
R2 0.162 0.162 0.162	R2	0.162	0.162	0.162
R2 Adj. 0.148 0.149	R2 Adj.	0.148	0.148	0.149
AIC 50497.8 50222.5 50349.5	AIC	50497.8	50222.5	50349.5
BIC 54028.1 53750.8 53879.1	BIC	54028.1	53750.8	53879.1
RMSE 0.44 0.44 0.44	RMSE	0.44	0.44	0.44
Std.Errors Heteroskedasticity- Heteroskedasticity- Heteroskedasticity- robust robust robust	Std.Errors	Heteroskedasticity- robust	Heteroskedasticity- robust	Heteroskedasticity- robust
FE: region Yes Yes Yes	FE: region	Yes	Yes	Yes

Table C.	8: Logit Regression	n Results Secondary	Education	(Interaction)

• +p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

	(M: 10 km)	(M:25 km)	(M:50 km)	(F: 10 km)	(F: 25 km)	(F: 50 km)
	Secondary Education	Secondary Education	Secondary Education	Secondary Education	Secondary Education	Secondary Education
Active	-0.5269	-0.3080+	0.2929***	0.3561	0.2614	0.2057**
	(0.4259)	(0.1747)	(0.0842)	(0.3648)	(0.1655)	(0.0776)
Inactive	0.1390	0.7979	0.1027	0.2627	0.6260	0.0252
	(0.7415)	(0.5804)	(0.1994)	(0.6977)	(0.4706)	(0.2078)
Religion	-0.3682***	-0.3710***	-0.3727***	-0.4425***	-0.4498***	-0.4476***
	(0.0548)	(0.0550)	(0.0548)	(0.0613)	(0.0616)	(0.0615)
Urban	-0.0112	-0.0090	-0.0124	-0.0145	-0.0202	-0.0138
	(0.0244)	(0.0246)	(0.0246)	(0.0245)	(0.0250)	(0.0247)
Drought	-2.7041**	-2.8446**	-2.7652**	-2.8071**	-2.7546**	-2.8575**
	(0.9391)	(0.9430)	(0.9418)	(1.0545)	(1.0572)	(1.0573)
Rain	0.0013**	0.0013**	0.0013**	0.0016***	0.0017***	0.0016***
	(0.0004)	(0.0004)	(0.0004)	(0.0005)	(0.0005)	(0.0005)
Agri. Land	0.0066**	0.0069**	0.0064**	0.0037	0.0035	0.0035
	(0.0024)	(0.0024)	(0.0024)	(0.0025)	(0.0025)	(0.0025)
Exp. Time	0.0676	0.0737**	0.0116	-0.0120	-0.0622**	-0.0051
	(0.0623)	(0.0263)	(0.0144)	(0.0514)	(0.0241)	(0.0139)
Age (18 - 26)	0.9863***	0.9750***	0.9281***	1.0682***	1.0678***	1.0390***
	(0.0618)	(0.0623)	(0.0629)	(0.0671)	(0.0676)	(0.0683)
Age (26 - 35)	0.4651***	0.4594***	0.4464***	0.6987***	0.7024***	0.6899***
	(0.0557)	(0.0559)	(0.0559)	(0.0614)	(0.0616)	(0.0616)
Age (36 -45)	0.4086***	0.4068***	0.4064***	0.4984***	0.4993***	0.4956***
	(0.0570)	(0.0571)	(0.0570)	(0.0629)	(0.0630)	(0.0630)
Age (56- 65)	-0.2559***	-0.2544***	-0.2523***	-0.5551***	-0.5576***	-0.5551***
	(0.0701)	(0.0702)	(0.0701)	(0.0882)	(0.0882)	(0.0882)
Age (66 +)	-0.7736***	-0.7711***	-0.7680***	-1.2729***	-1.2761***	-1.2721***
	(0.0888)	(0.0890)	(0.0889)	(0.1323)	(0.1323)	(0.1325)
Num.Ob s.	21369	21273	21331	21754	21635	21707
R2	0.155	0.156	0.156	0.184	0.184	0.184
R2 Adj.	0.128	0.129	0.129	0.157	0.157	0.157
AIC	25776.0	25637.2	25707.1	24804.2	24651.9	24728.7
BIC	28979.8	28839.2	28910.2	27991.3	27836.7	27914.9
RMSE	0.45	0.45	0.45	0.43	0.43	0.43
Std.Erro rs	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust
FE: region	Yes	Yes	Yes	Yes	Yes	Yes

Table C. 9: Logit Regression Results Secondary Education (Gender Differentiated)

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. M = Male. F = Female.

	(10 km) Education	(25 km) Education	(50 km) Education
Active	0.4347*	0.1948*	0.0665
	(0.2114)	(0.0867)	(0.0432)
Inactive	0.5769+	0.5961**	0.4420***
	(0.3255)	(0.2189)	(0.1199)
Gender	0.5377***	0.5375***	0.5374***
	(0.0173)	(0.0173)	(0.0173)
Ethnicity	1.0556***	1.0554***	1.0551***
	(0.1125)	(0.1125)	(0.1126)
Urban	0.0591***	0.0585***	0.0590***
	(0.0157)	(0.0160)	(0.0158)
Drought	-1.3605*	-1.3527*	-1.2994*
	(0.5347)	(0.5361)	(0.5352)
Rain	0.0003	0.0004	0.0004
	(0.0003)	(0.0003)	(0.0003)
Agri. Land	-0.0002	-0.0003	-0.0004
	(0.0015)	(0.0015)	(0.0015)
Exp. Time	-0.0171	-0.0278*	-0.0086
	(0.0292)	(0.0126)	(0.0075)
Age (18 - 25)	0.8154***	0.8114***	0.8046***
	(0.0359)	(0.0362)	(0.0364)
Age (26 - 35)	0.6951***	0.6938***	0.6909***
	(0.0331)	(0.0331)	(0.0331)
Age (36 - 45)	0.3580***	0.3570***	0.3563***
	(0.0333)	(0.0333)	(0.0333)
Age (56 - 65)	-0.4007***	-0.4016***	-0.4008***
	(0.0419)	(0.0419)	(0.0419)
Age (66 +)	-0.8980***	-0.8986***	-0.8989***
	(0.0503)	(0.0504)	(0.0503)
Num.Obs.	43317	43102	43232
R2	0.312	0.312	0.312
R2 Adj.	0.305	0.306	0.306
AIC	173858.7	173041.7	173542.8
BIC	177416.0	176596.9	177099.2
RMSE	1.78	1.78	1.78
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table C. 10: Regression Results Ethnicity

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. Ethnicity = 1, if the respondent is of America n, Chinese or European descent and 0, if not. Ethnicity = 0 is the reference category.

F – Test (25 km): F = 3.39, p < 0.1

F – Test (50 km): F = 9.41, p < 0.01

	(10 km) Education	(25 km) Education	(50 km) Education
Active	0.5439*	0.2352*	0.0515
	(0.2334)	(0.1001)	(0.0528)
Inactive	0.5706+	0.5963**	0.4420***
	(0.3262)	(0.2189)	(0.1199)
Gender	0.5387***	0.5391***	0.5353***
	(0.0173)	(0.0175)	(0.0180)
Ethnicity	1.0556***	1.0551***	1.0551***
-	(0.1125)	(0.1125)	(0.1125)
Urban	0.0591***	0.0585***	0.0589***
	(0.0157)	(0.0160)	(0.0158)
Drought	-1.3610*	-1.3510*	-1.3021*
-	(0.5348)	(0.5361)	(0.5353)
Rain	0.0003	0.0004	0.0004
	(0.0003)	(0.0003)	(0.0003)
Agri. Land	-0.0002	-0.0003	-0.0004
	(0.0015)	(0.0015)	(0.0015)
Exp. Time	-0.0155	-0.0279*	-0.0086
	(0.0296)	(0.0126)	(0.0075)
Age (18 - 25)	0.8153***	0.8115***	0.8046***
	(0.0359)	(0.0362)	(0.0364)
Age (26 - 36)	0.6952***	0.6938***	0.6909***
	(0.0331)	(0.0331)	(0.0331)
Age (36 - 45)	0.3580***	0.3570***	0.3563***
	(0.0333)	(0.0333)	(0.0333)
Age_56_65	-0.4007***	-0.4017***	-0.4007***
	(0.0419)	(0.0419)	(0.0419)
Age (66 +)	-0.8982***	-0.8989***	-0.8986***
	(0.0503)	(0.0504)	(0.0503)
Active × Gender	-0.2567	-0.0845	0.0324
	(0.2606)	(0.1102)	(0.0628)
Num.Obs.	43317	43102	43232
R2	0.312	0.312	0.312
R2 Adj.	0.305	0.306	0.306
AIC	173859.9	173043.2	173544.5
BIC	177425.8	176607.2	177109.7
RMSE	1.78	1.78	1.78
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table C.11: Regression Results Ethnicity (Interaction)

 + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. Ethnicity = 1, if the respondent is of American, Chinese or European descent and 0, if not. Ethnicity = 0 is the reference category.

	(M:10 km)	(M:25 km)	(M: 50 km)	(F:10 km)	(F: 25 km)	(F: 50 km)
	Education	Education	Education	Education	Education	Educatio n
Active	0.4410	0.2222+	0.1171+	0.3966	0.1431	0.0156
	(0.3333)	(0.1227)	(0.0613)	(0.2732)	(0.1238)	(0.0606)
Inactive	0.6633	0.5611*	0.2203	0.6353	0.7150*	0.6195***
	(0.4362)	(0.2658)	(0.1603)	(0.4826)	(0.3574)	(0.1750)
Ethnicit	1.0035***	1.0034***	1.0011***	1.0860***	1.0856***	1.0855***
У						
	(0.1608)	(0.1608)	(0.1609)	(0.1542)	(0.1542)	(0.1541)
Urban	0.0493*	0.0477*	0.0478*	0.0691**	0.0690**	0.0700**
	(0.0229)	(0.0232)	(0.0230)	(0.0214)	(0.0220)	(0.0214)
Drought	-1.4810+	-1.4927*	-1.4536+	-0.9581	-0.9276	-0.8552
	(0.7583)	(0.7602)	(0.7598)	(0.7551)	(0.7572)	(0.7553)
Rain	0.0000	0.0000	0.0000	0.0004	0.0004	0.0004
	(0.0004)	(0.0004)	(0.0004)	(0.0004)	(0.0004)	(0.0004)
Agri. Land	-0.0007	-0.0007	-0.0008	0.0006	0.0004	0.0003
	(0.0021)	(0.0021)	(0.0021)	(0.0021)	(0.0021)	(0.0021)
Exp- Time	-0.0337	-0.0264+	-0.0101	0.0031	-0.0285	-0.0070
	(0.0409)	(0.0157)	(0.0102)	(0.0418)	(0.0197)	(0.0110)
Age (18 - 25)	0.6982***	0.6954***	0.6862***	0.9638***	0.9608***	0.9572***
	(0.0523)	(0.0527)	(0.0530)	(0.0497)	(0.0500)	(0.0504)
Age (26 - 35)	0.6078***	0.6054***	0.6032***	0.7962***	0.7972***	0.7931***
	(0.0481)	(0.0482)	(0.0482)	(0.0455)	(0.0456)	(0.0456)
Age (36 - 45)	0.3244***	0.3228***	0.3232***	0.4073***	0.4072***	0.4056***
	(0.0479)	(0.0479)	(0.0479)	(0.0464)	(0.0464)	(0.0464)
Age (56 - 65)	-0.3296***	-0.3303***	-0.3292***	-0.5356***	-0.5366***	-0.5366***
	(0.0578)	(0.0578)	(0.0578)	(0.0605)	(0.0605)	(0.0605)
Age (66+)	-0.8611***	-0.8608***	-0.8599***	-1.0527***	-1.0544***	-1.0557***
	(0.0673)	(0.0674)	(0.0674)	(0.0764)	(0.0765)	(0.0764)
Num.Ob s.	21448	21352	21410	21869	21750	21822
R2	0.287	0.287	0.287	0.350	0.351	0.351
R2 Adj.	0.273	0.274	0.273	0.338	0.339	0.339
AIC	86851.8	86493.8	86721.3	86813.3	86354.6	86622.1
BIC	90112.9	89753.1	89981.7	90082.4	89621.4	89890.3
RMSE	1.80	1.80	1.80	1.73	1.73	1.73
Std.Erro rs	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust
FE:	Yes	Yes	Yes	Yes	Yes	Yes
region						

Table C. 12: Regression Results Ethnicity (Gender Differentiated)

• p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. M = Male. F = Female.

	Education	Education	Education
Active	0.4062+	0.2383**	0.0882*
	(0.2082)	(0.0863)	(0.0427)
Inactive	0.4412	0.5197*	0.4133***
	(0.3285)	(0.2182)	(0.1172)
Gender	0.5421***	0.5421***	0.5418***
	(0.0169)	(0.0170)	(0.0169)
Religion	-0.4696***	-0.4754***	-0.4726***
	(0.0309)	(0.0309)	(0.0309)
Urban	0.8915***	0.8934***	0.8928***
	(0.0223)	(0.0224)	(0.0224)
Drought	-0.9329+	-0.9154+	-0.8772+
	(0.5241)	(0.5254)	(0.5245)
Rain	0.0003	0.0003	0.0003
	(0.0002)	(0.0002)	(0.0002)
Agri. Land	-0.0005	-0.0005	-0.0007
	(0.0014)	(0.0014)	(0.0014)
Exp. Time	-0.0200	-0.0319*	-0.0082
	(0.0292)	(0.0124)	(0.0074)
Age (18 - 25)	0.8191***	0.8130***	0.8052***
	(0.0340)	(0.0343)	(0.0345)
Age (26 - 35)	0.6830***	0.6817***	0.6777***
	(0.0316)	(0.0316)	(0.0316)
Age (36 - 45)	0.3468***	0.3462***	0.3449***
	(0.0326)	(0.0326)	(0.0326)
Age (56 - 65)	-0.3865***	-0.3871***	-0.3864***
	(0.0413)	(0.0413)	(0.0413)
Age (66 +)	-0.8822***	-0.8826***	-0.8826***
	(0.0496)	(0.0496)	(0.0496)
Num.Obs.	43317	43102	43232
R2	0.341	0.341	0.341
R2 Adj.	0.334	0.335	0.335
AIC	172010.9	171196.2	171692.8
BIC	175568.2	174751.5	175249.2
RMSE	1.75	1.75	1.75
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table C.13: Regression Results Urban

+ p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. Urban = 1 if the respondent lives in an urban enumeration area. Urban = 0, if the respondent lives in a rural enumeration area. Urban = 0 is the reference category. M = Male. F = Female.

F – Test (50 km): F = 7.37, p < 0.01

	Education	Education	Education
Active	0.5306*	0.2810**	0.0723
	(0.2248)	(0.0990)	(0.0522)
Inactive	0.4341	0.5199*	0.4133***
	(0.3293)	(0.2182)	(0.1172)
Gender	0.5432***	0.5438***	0.5395***
	(0.0169)	(0.0172)	(0.0176)
Religion	-0.4696***	-0.4753***	-0.4726***
	(0.0309)	(0.0309)	(0.0309)
Urban	0.8916***	0.8934***	0.8928***
	(0.0223)	(0.0224)	(0.0224)
Drought	-0.9335+	-0.9136+	-0.8800+
	(0.5241)	(0.5254)	(0.5246)
Rain	0.0003	0.0003	0.0003
	(0.0002)	(0.0002)	(0.0002)
Agri. Land	-0.0005	-0.0005	-0.0007
	(0.0014)	(0.0014)	(0.0014)
Exp. Time	-0.0182	-0.0320**	-0.0082
	(0.0297)	(0.0124)	(0.0074)
Age (18 - 25)	0.8190***	0.8131***	0.8052***
	(0.0340)	(0.0343)	(0.0345)
Age (26 - 35)	0.6831***	0.6817***	0.6776***
	(0.0316)	(0.0316)	(0.0316)
Age (36 - 45)	0.3468***	0.3462***	0.3449***
	(0.0326)	(0.0326)	(0.0326)
Age (56 - 65)	-0.3866***	-0.3872***	-0.3862***
	(0.0413)	(0.0413)	(0.0413)
Age (66+)	-0.8825***	-0.8828***	-0.8823***
	(0.0496)	(0.0496)	(0.0496)
Active × Gender	-0.2926	-0.0893	0.0342
	(0.2662)	(0.1096)	(0.0622)
Num.Obs.	43317	43102	43232
R2	0.341	0.341	0.341
R2 Adj.	0.334	0.335	0.335
AIC	172011.8	171197.7	171694.5
BIC	175577.7	174761.6	175259.7
RMSE	1.75	1.75	1.75
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table C. 14: Regression Results Urban (Interaction)

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. Urban = 1 if the respondent lives in an urban enumeration area. Urban = 0, if the respondent lives in a rural enumeration area. Urban = 0 is the reference category. M = Male. F = female.

	(M:10 km)	(M: 25 km)	(M: 50 km)	(F: 10 km)	(F: 25 km)	(F: 50 km)
	Education	Education	Education	Education	Education	Education
Active	0.4435	0.2676*	0.1451*	0.3492	0.1850	0.0317
	(0.3478)	(0.1239)	(0.0607)	(0.2576)	(0.1218)	(0.0599)
Inactive	0.5318	0.4803+	0.2052	0.4951	0.6420+	0.5788***
	(0.4211)	(0.2609)	(0.1575)	(0.4999)	(0.3562)	(0.1705)
Religion	-0.4886***	-0.4917***	-0.4918***	-0.4805***	-0.4887***	-0.4826***
	(0.0441)	(0.0442)	(0.0441)	(0.0431)	(0.0431)	(0.0431)
Urban	0.9355***	0.9379***	0.9362***	0.8564***	0.8571***	0.8580***
	(0.0321)	(0.0322)	(0.0321)	(0.0307)	(0.0309)	(0.0308)
Drought	-1.0172	-1.0111	-0.9921	-0.5428	-0.5099	-0.4497
	(0.7423)	(0.7441)	(0.7439)	(0.7429)	(0.7448)	(0.7429)
Rain	0.0000	0.0000	0.0000	0.0002	0.0002	0.0002
	(0.0004)	(0.0004)	(0.0004)	(0.0003)	(0.0003)	(0.0003)
Agri. Land	-0.0012	-0.0012	-0.0014	0.0006	0.0004	0.0003
_	(0.0020)	(0.0020)	(0.0020)	(0.0020)	(0.0020)	(0.0020)
Exp. Time	-0.0467	-0.0310*	-0.0102	0.0080	-0.0321+	-0.0060
	(0.0434)	(0.0158)	(0.0101)	(0.0411)	(0.0191)	(0.0108)
Age (18 -25)	0.6744***	0.6694***	0.6580***	0.9933***	0.9882***	0.9848***
	(0.0491)	(0.0495)	(0.0499)	(0.0473)	(0.0476)	(0.0481)
Age (26 -35)	0.5683***	0.5659***	0.5619***	0.8087***	0.8097***	0.8051***
	(0.0457)	(0.0458)	(0.0458)	(0.0436)	(0.0437)	(0.0437)
Age (36 -45)	0.2998***	0.2987***	0.2982***	0.4087***	0.4088***	0.4070***
	(0.0467)	(0.0468)	(0.0467)	(0.0455)	(0.0455)	(0.0455)
Age (56 -65)	-0.3184***	-0.3186***	-0.3174***	-0.5208***	-0.5214***	-0.5216***
	(0.0566)	(0.0566)	(0.0566)	(0.0599)	(0.0600)	(0.0600)
Age (66+)	-0.8579***	-0.8571***	-0.8561***	-1.0249***	-1.0268***	-1.0274***
	(0.0662)	(0.0663)	(0.0663)	(0.0749)	(0.0751)	(0.0750)
Num.Ob s.	21448	21352	21410	21869	21750	21822
R2	0.320	0.320	0.320	0.377	0.378	0.378
R2 Adj.	0.306	0.307	0.306	0.365	0.366	0.366
AIC	85850.8	85494.2	85719.9	85896.3	85439.3	85703.5
BIC	89111.9	88753.5	88980.3	89165.4	88706.1	88971.7
RMSE	1.76	1.76	1.76	1.69	1.69	1.69
Std.Erro rs	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust	Heteroskedastici ty-robust
FE: region	Yes	Yes	Yes	Yes	Yes	Yes

Table C.15: Regression Results Urban (Gender Differentiated)

+ p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001. Urban = 1 if the respondent lives in an urban enumeration area. Urban = 0, if the respondent lives in a rural enumeration area. Urban = 0 is the reference category. M = Male. F = female.

•

Appendix D: Testing the Causal Mechanisms

To test the causal mechanisms, I operationalized *active* and *inactive* according to Wegenast et al. (2022), without considering the age of respondents and the timing of the LSLAs, as considering these is not necessary when the dependent variable is not education. Therefore, *active* is coded as 1 for those within the buffer of an active LSLA, and *inactive* for those within the buffer of an inactive LSLA. The reference group consists of individuals not located in any LSLA buffer. I removed respondents who are in the buffer of both an active and an inactive LSLA. For the operationalization of control variables, I exclusively used variables from the Afrobarometer dataset. Table D.1 details the operationalization of all variables used in my analysis. the regression models, the reference category for binary variables is always the group that is assigned a value of 0.

Variable	Operationalization
	1 = The respondent is within the buffer of an active
	LSLA.
Active	
	0 = The respondent is not within the buffer of an
	active LSLA.
	1 = The respondent is within the buffer of an inactive
	LSLA.
Inactive	
	0 = The respondent is not within the buffer of an
	inactive LSLA.
	1 = There is a school in the primary enumeration
	area of the respondent.
School	
	0 = There is no school in the enumeration area of the
	respondent.
	1 = The respondent lives in an urban enumeration
Urban	area.
	0 = The respondent lives in a rural enumeration area.

Table D.1:	Operationalization	of Variables
TUDIC D.I.	operationalization	oj variabies

	1 = The respondent and his or her family felt	
	insecure several times, many times, or always while	
	walking in their neighborhood in the past year.	
Insecurity		
	0 = The respondent and his or her family felt	
	insecure never, or just once or twice while walking in	
	their neighborhood in the past year.	
	1 = There is an electricity grid in the enumeration	
	area of the respondent.	
Electricity		
	0 = There is no electricity grid in the enumeration	
	area of the respondent.	
	1 = There is a market stall in the enumeration area of	
	the respondent.	
Market Stall		
	0 = There is no market stall in the enumeration area	
	of the respondent.	
Contra	1 = Male	
Genaer	0 - Fomalo	
	How often does the bousehold in which the	
	respondent lives receive remittances from abroad2.0	
Pomittancos	= nover 1 = Less than once a voar 2 = At least once	
Kenntunces	- Hever, $1 -$ Less than once a year, $2 -$ At least once	
	a year, $3 = At least every six months, 4 = At least$	
	every three months, 5 = At least once a month	
	I = The respondent or anyone in the family of the	
	respondent did not have enough to eat several	
Food inconvrity	times, many times, or always in the past year.	
rooumsecunty	Ω = The respondent or anyone in the family of the	
	respondent did not have enough to eat never, or just	
	once or twice in the past year	
	1 = The respondent or anyone in his family went	
	without cash income many times several times or	
	always in the past year	
No income		
	0 = The respondent or anyone in his family went	
	without cash income never, or just once or twice in	
	the past year.	

	1 = The respondent is an active member or a leader	
	of a community organization.	
Community		
	0 = The respondent is not a member or is an inactive	
	member of a community organization	
Age	Age of the Respondent at the time of the survey.	
Education	Operationalized like in the main text (Chapter 4.2).	
Household	Number of household members.	

To enhance confidence in my theoretical mechanisms, I tested if LSLAs negatively impact school infrastructure, thus affecting the supply side of education. I conducted a logistic regression and controlled for whether the respondent lives in an urban or rural area, as infrastructure tends to be better in urban areas of developing countries (Hlalele, 2014). Moreover, I also controlled for neighborhood security, which can affect both infrastructure and the placement of LSLAs (Balestri & Maggioni, 2019). Additionally, I used access to electricity as a proxy for state capacity, which can influence infrastructure (Koren & Sarbahi, 2018).

The results of the logistic regressions are shown in Table D.2, indicating that LSLAs tend to improve school infrastructure. Looking at the school infrastructure within a 10-kilometer buffer of the LSLAs, there is a statistically significant positive effect on school infrastructure (F = 30.882, p < 0.001). This can potentially be attributed to corporate social responsibility programs by international investors. This finding boosts confidence in my theory, as it indicates that the negative impact of LSLAs on education does not stem from a deterioration of the education supply side. However, the results should be interpreted with caution, as they do not reflect the influence of LSLAs on school infrastructure during the period when the respondents were between 0 and 16 years old.

	(10 km)	(25 km)	(50 km)
	School	School	School
Active	0.32*	-0.11	-0.07
	(0.14)	(0.07)	(0.06)
Inactive	-0.86***	-1.63***	-1.03***
	(0.16)	(0.13)	(0.11)
Urban	0.16***	0.16**	0.16**
	(0.05)	(0.05)	(0.05)
Insecurity	0.01	0.01	0.01
	(0.04)	(0.04)	(0.04)
Electricity	1.42***	1.44***	1.45***
	(0.05)	(0.05)	(0.05)
Num.Obs.	30877	30805	29831
R2	0.171	0.175	0.173
R2 Adj.	0.159	0.162	0.160
AIC	25363.1	25224.6	24701.2
BIC	26972.3	26833.4	26287.1
RMSE	0.36	0.36	0.36
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table D.2 : Logit Regression Results School Infrastructure

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

Next, I examined whether LSLAs have a statistically significant negative impact on food insecurity and income. The results are presented in tables D.3 and D.4. In both regressions, I controlled for several factors including whether the respondent lives in an urban area, where food insecurity is lower and income is higher (De Magalhães, 2018; Bjornlund et al., 2022), whether the respondent receives remittances which can generate additional income and thereby reduce food insecurity, whether the respondent is a member of a community association which can provide financial support and mitigate food insecurity (Fisher & Lewin, 2013), whether the respondent has a market stall nearby, used as a proxy for economic activity, the number of people in the household, as larger households may leave fewer resources per person (Biyase & Zwane, 2018), and whether the respondent lives in an insecure area, which can diminish income prospects and exacerbate food insecurity (Justino, 2012; Kemmerling et al., 2022). Additionally, I considered gender differences, since men in developing countries often have higher incomes than women and are less frequently affected by food insecurity (Broussard, 2020; Kim, 2020). Additionally, I accounted for age, as older

respondents may no longer be able to work, which potentially affects their ability to reduce food insecurity.

	(10 km)	(25 km)	(50 km)
	No Income	No Income	No Income
Active	-0.027	-0.021	0.047
	(0.088)	(0.047)	(0.044)
Inactive	0.274	-0.193	-0.223*
	(0.194)	(0.123)	(0.105)
Age	0.005***	0.005***	0.005***
	(0.001)	(0.001)	(0.001)
Urban	-0.492***	-0.492***	-0.492***
	(0.029)	(0.029)	(0.029)
Gender	-0.076***	-0.075***	-0.076***
	(0.023)	(0.023)	(0.023)
Market	-0.096**	-0.093**	-0.091**
	(0.031)	(0.031)	(0.031)
Remittances	-0.132***	-0.131***	-0.129***
	(0.008)	(0.008)	(0.008)
Community	-0.081**	-0.082**	-0.078**
	(0.028)	(0.028)	(0.029)
Household	0.016**	0.016**	0.015**
	(0.005)	(0.005)	(0.006)
Insecurity	0.581***	0.582***	0.583***
	(0.029)	(0.029)	(0.029)
Num.Obs.	45035	44966	43962
R2	0.194	0.194	0.196
R2 Adj.	0.180	0.180	0.182
AIC	48147.6	48071.3	46894.9
BIC	51677.2	51600.3	50406.1
RMSE	0.42	0.42	0.42
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table D.3 : Logit Regression Results No Income

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

The results indicate that LSLAs have neither a statistically significant negative impact on food insecurity nor on income, which weakens confidence in my theoretical mechanisms.

	(10 km)	(25 km)	(50 km)
	Food Insecurity	Food Insecurity	Food Insecurity
Active	-0.095	-0.049	-0.046
	(0.090)	(0.047)	(0.042)
Inactive	-0.346	-0.018	-0.109
	(0.200)	(0.119)	(0.097)
Age	0.011***	0.011***	0.011***
	(0.001)	(0.001)	(0.001)
Urban	-0.410***	-0.411***	-0.411***
	(0.030)	(0.030)	(0.030)
Gender	-0.117***	-0.116***	-0.115***
	(0.022)	(0.022)	(0.023)
Market	-0.093**	-0.094**	-0.096**
	(0.030)	(0.030)	(0.030)
Remittances	-0.123***	-0.123***	-0.120***
	(0.009)	(0.009)	(0.009)
Community	-0.098***	-0.099***	-0.099***
	(0.027)	(0.027)	(0.028)
Household	0.008	0.008	0.006
	(0.005)	(0.005)	(0.005)
Insecurity	0.640***	0.639***	0.636***
	(0.026)	(0.026)	(0.027)
Num.Obs.	44880	44811	43807
R2	0.137	0.136	0.138
R2 Adj.	0.122	0.122	0.123
AIC	49183.5	49120.2	48026.8
BIC	52676.9	52613.0	51501.8
RMSE	0.43	0.43	0.43
Std.Errors	Heteroskedasticity-robust	Heteroskedasticity-robust	Heteroskedasticity-robust
FE: region	Yes	Yes	Yes

Table D. 4: Logit Regression Results Food Insecurity

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

Finally, I examined the influence of food insecurity and income on education. The results, displayed in Table D.5, indicate that both lack of income and food insecurity, as expected, have a statistically significant negative impact on education. I used the same control variables because, as outlined in my theoretical framework, food insecurity, income, and education are closely interconnected. I reported the results for the different datasets corresponding to each buffer size used, but in this case, buffer size is not relevant because I am not using the active and inactive variables.

	(10 km)	(25 km)	(50 km)
	Education	Education	Education
Food Insecurity	-0.420***	-0.422***	-0.418***
	(0.019)	(0.019)	(0.020)
No Income	-0.389***	-0.389***	-0.385***
	(0.021)	(0.021)	(0.021)
Age	-0.034***	-0.034***	-0.034***
	(0.001)	(0.001)	(0.001)
Urban	0.571***	0.569***	0.565***
	(0.024)	(0.024)	(0.024)
Gender	0.520***	0.520***	0.512***
	(0.017)	(0.017)	(0.017)
Market	0.113***	0.112***	0.111***
	(0.022)	(0.023)	(0.023)
Electricity	0.613***	0.615***	0.619***
	(0.026)	(0.026)	(0.026)
Community	0.408***	0.408***	0.408***
	(0.021)	(0.021)	(0.021)
Household	0.008*	0.008*	0.008*
	(0.004)	(0.004)	(0.004)
Insecurity	0.052*	0.051*	0.061**
	(0.020)	(0.020)	(0.020)
Num.Obs.	45770	45699	44679
R2	0.365	0.365	0.367
R2 Adj.	0.359	0.359	0.361
AIC	181919.9	181638.0	177471.1
BIC	185578.4	185295.8	181110.7
RMSE	1.75	1.75	1.75
Std.Errors	Heteroskedasticity- robust	Heteroskedasticity- robust	Heteroskedasticity- robust
FE: region	Yes	Yes	Yes

Table D.5: Regression Results Causal Mechanisms

• + p < 0.1, * p < 0.05, ** p < 0.01, *** p < 0.001

Overall, it can be said that the results of my causal mechanism testing are mixed and not sufficient to increase confidence in my theory. The causal mechanisms tested here also form the basis for my theory regarding the gendered impacts of LSLAs on education. However, I am unable to test changes in the gender-specific educational preferences of households. This implies that households are a black box, preventing me from investigating how LSLAs influence household decisions on education.

References Appendix

- Balestri, S., & Maggioni, M. A. (2021). This Land Is My Land! Large-Scale Land Acquisitions and Conflict Events in Sub-Saharan Africa. *Defence and Peace Economics*, 32(4), 427– 450. https://doi.org/10.1080/10242694.2019.1647727
- Biyase, M., & Zwane, T. (2018). An Empirical Analysis of the Determinants of Poverty and Household Welfare in South Africa. *The Journal of Developing Areas*, 52(1), 115–130. https://doi.org/10.1353/jda.2018.0008
- Bjornlund, V., Bjornlund, H., & van Rooyen, A. (2022). Why food insecurity persists in sub-Saharan Africa: A review of existing evidence. *Food Security*, *14*(4), 845–864. https://doi.org/10.1007/s12571-022-01256-1
- Broussard, N. H. (2019). What explains gender differences in food insecurity? *Food Policy*, *83*, 180–194. https://doi.org/10.1016/j.foodpol.2019.01.003
- De Magalhães, L., & Santaeulàlia-Llopis, R. (2018). The consumption, income, and wealth of the poorest: An empirical analysis of economic inequality in rural and urban Sub-Saharan Africa for macroeconomists." *Journal of Development Economics* 134 (2018): 350-371.https://www.sciencedirect.com/science/article/pii/s0304387818305017
- Fisher, M., & Lewin, P. A. (2013). Household, community, and policy determinants of food insecurity in rural Malawi. *Development Southern Africa*, 30(4-05), 451–467. https://doi.org/10.1080/0376835X.2013.830966
- Hlalele, D. (2014). Rural Education in South Africa: Concepts and Practices. *Mediterranean Journal of Social Sciences.* Advance online publication. https://doi.org/10.5901/mjss.2014.v5n4p462
- Justino, P. (2012). War and Poverty. *IDS Working Papers*, 2012(391), 1–29. https://doi.org/10.1111/j.2040-0209.2012.00391.x
- Kemmerling, B., Schetter, C., & Wirkus, L. (2022). The logics of war and food (in)security. *Global Food Security*, *33*, 100634. https://doi.org/10.1016/j.gfs.2022.100634
- Kim, S.-B. (2020). Gender earnings gap among the youth in Malawi. *African Development Review*, *32*(2), 176–187. https://doi.org/10.1111/1467-8268.12426
- Koren, O., & Sarbahi, A. K. (2018). State Capacity, Insurgency, and Civil War: A Disaggregated Analysis. International Studies Quarterly, 62(2), 274–288. https://doi.org/10.1093/isq/sqx076
- Wegenast, T., Richetta, C., Krauser, M., & Leibik, A. (2022). Grabbed trust? The impact of large-scale land acquisitions on social trust in Africa. *World Development*, 159, 106038. https://doi.org/10.1016/j.worlddev.2022.106038



Previous Munich Papers in Political Economy:

2020

Betz, Timm and Amy Pond. "Political Ownership", MPPE No. 1/2020, Munich. (ISSN)2701-3456

- Chatziathanasiou, Konstantin, Hippel, Svenja and Michael Kurschilgen. "Property, Redistribution, and the Status Quo. A laboratory study", MPPE No. 2/2020, Munich. (ISSN)2701-3456
- Becker, Annette, Hottenrott, Hanna and Anwesha Mukherjee. "Division of Labor in R&D? Firm Size and Specialization in Corporate Research", MPPE No. 3/2020, Munich. (ISSN)2701-3456
- Steinert, Janina Isabel, Satish, Rucha Vasumati, Stips, Felix and Sebastian Vollmer. "Commitment or Concealment? Impacts and Use of a Portable Saving Device: Evidence from a Field Experiment in Urban India, MPPE No. 4/2020, Munich. (ISSN)2701-3456
- Messerschmidt, Luca and Nicole Janz. "Unravelling the `race to the bottom' argument: How does FDI affect different types of labour rights?", MPPE No. 5/2020, Munich. (ISSN)2701-3456.
- Chowdhury, Subhasish M., Esteve-González, Patricia and Anwesha Mukherjee. "Heterogeneity, Leveling the Playing Field, and Affirmative Action in Contests", MPPE No. 6/2020, Munich. (ISSN)2701-3456
- Drobner, Christoph. "Motivated Beliefs and Anticipation of Uncertainty Resolution", MPPE No. 7/2020, Munich. (ISSN)2701-3456
- Chatziathanasiou, Konstantin, Hippel, Svenja and Michael Kurschilgen. "Do rights to resistance discipline the elites? An experiment on the threat of overthrow", MPPE No. 8/2020, Munich. (ISSN)2701-3456
- Siddique, Abu, Rahman, Tabassum, Pakrashi, Debayan, Islam, Asad, and Firoz Ahmed. "Raising COVID-19 Awareness in Rural Communities: A Randomized Experiment in Bangladesh and India", MPPE No. 9/2020, Munich. (ISSN)2701-3456

2021

- Siddique, Abu. "Behavioral Consequences of Religious Education", MPPE No. 01/2021, Munich. (ISSN)2701-3456
- Vlassopoulos, Michael, Siddique, Abu, Rahman, Tabassum, Pakrashi, Debayan, Islam, Asad, and Firoz Ahmed. "Improving Women's Mental Health During a Pandemic", MPPE No. 02/2021, Munich. (ISSN)2701-3456
- March, Christoph, Schieferdecker, Ina."Technological Sovereignty as Ability, not Autarky", MPPE No. 03/2021, Munich. (ISSN)2701-3456
- Hassan, Hashibul, Islam, Asad, Siddique, Abu, and Liang Choon Wang. "Telementoring and homeschooling during school closures: A randomized experiment in rural Bangladesh", MPPE No. 04/2021, Munich. (ISSN)2701-3456
- Angerer, Silvia, Dutcher ,Glenn, Glätzle-Rützler, Daniela, Lergetporer, Philipp, and Matthias Sutter. "The formation of risk preferences through small-scale events", MPPE No. 05/2021, Munich. (ISSN)2701-3456
- Hermes, Henning, Lergetporer, Philipp, Peter, Frauke and Simon Wiederhold. "Behavioral Barriers and the Socioeconomic Gap in Child Care Enrollment", MPPE No. 06/2021, Munich. (ISSN)2701-3456
- Schwierzy, Julian. "Digitalisation of Production: Industrial Additive Manufacturing and its Implications for Competition and Social Welfare", MPPE No. 07/2021, Munich. (ISSN)2701-3456

Kurschilgen, Michael. "Moral awareness polarizes people's fairness judgments", MPPE No. 08/2021, Munich. (ISSN)2701-3456

Drobner, Christoph, and Sebastian J. Goerg. "Motivated Belief Updating and Rationalization of Information", MPPE No. 09/2021, Munich. (ISSN)2701-3456



Previous Munich Papers in Political Economy:

2022

- Lergetporer, Philipp, and Ludger Woessmann. "Income Contingency and the Electorate's Support for Tuition", MPPE No. 01/2022, Munich. (ISSN)2701-3456
- Angerer, Silvia, Glätzle-Rützler, Daniela, Lergetporer, Philipp, and Thomas Rittmannsberger. "How does the vaccine approval procedure affect COVID-19 vaccination intentions?", MPPE No. 02/2022, Munich. (ISSN)2701-3456
- Antonioli, Davide, Alberto Marzucchi, Francesco Rentocchini, and Simone Vannuccini. "Robot Adoption and Innovation Activities", MPPE No. 03/2022, Munich. (ISSN)2701-3456
- Hoeft, Leonard, Michael Kurschilgen, and Wladislaw Mill. "Norms as Obligations", MPPE No. 04/2022, Munich. (ISSN)2701-3456
- Stöhr, Valentina. "Climate protection in Germany: Party cues in a multi-party system", MPPE No. 05/2022, Munich. (ISSN)2701-3456
- Schönmann, Manuela, Anja Bodenschatz, Matthias Uhl, and Gari Walkowitz. "The Care-Dependent are Less Averse to Care Robots: Comparing Intuitions of the Affected and the Non-Affected", MPPE No. 06/2022, Munich. (ISSN)2701-3456
- Hermes, Henning, Marina Krauß, Philipp Lergetporer, Frauke Peter, and Simon Wiederhold. "Early Child Care and Labor Supply of Lower-SES Mothers: A Randomized Controlled Trial", MPPE No. 07/2022, Munich. (ISSN)2701-3456

2023

- Angerer, Silvia, Glätzle-Rützler, Daniela, Lergetporer, Philipp, and Thomas Rittmannsberger. "Beliefs about social norms and (the polarization of) COVID-19 vaccination readiness", MPPE No. 01/2023, Munich. (ISSN)2701-3456
- Blesse, Sebastian, Lergetporer, Philipp, Nover, Justus, and Katharina Werner. "Transparency and Policy Competition: Experimental Evidence from German Citizens and Politicians", MPPE No. 02/2023, Munich. (ISSN)2701-3456
- Sternberg, Henrike, Steinert, Janina Isabel, and Tim Büthe. "Compliance in the Public versus the Private Realm: Economic Preferences, Institutional Trust and COVID-19 Health Behaviors ", MPPE No. 03/2023, Munich. (ISSN)2701-3456
- Cantner, Fabienne, Christoph Drobner, and Sebastian J. Goerg. "Greenwashing your personality", MPPE No. 04/2023, Munich. (ISSN)2701-3456
- Goerg, Sebastian, Pondorfer, Andreas, and Valentina Stöhr. "Public support for more ambitious climate policies", MPPE No. 05/2023, Munich. (ISSN)2701-3456
- Fischer, Mira, Elisabeth Grewenig, Philipp Lergetporer, Katharina Werner, and Helen Zeidler. "The E-Word On the Public Acceptance of Experiments", MPPE No. 06/2023, Munich. (ISSN)2701-3456
- Lergetporer, Philipp, Katharina Wedel, and Katharina Werner. "Automatability of occupations, workers' labor-market expectations, and willingness to train", MPPE No. 07/2023, Munich. (ISSN)2701-3456



Previous Munich Papers in Political Economy:

2024

- Hermes, Henning, Philipp Lergetporer, Fabian Mierisch, Guido Schwerdt, and Simon Wiederhold. "Does Information about Inequality and Discrimination in Early Child Care Affect Policy Preferences?", MPPE No. 01/2024, Munich. (ISSN)2701-3456
- Etzerodt, Søren Frank, and Niels Jørgen Mau Pedersen. "Formula-based Grants as Pork Barrel Politics: Targetability and the Politicalstrategic Use of Grants", MPPE No. 02/2024, Munich. (ISSN)2701-3456
- Angerer, Silvia, Hanna Brosch, Daniela Glätzle-Rützler, Philipp Lergetporer, and Thomas Rittmannsberger. "Discrimination in the general population", MPPE No. 03/2024, Munich. (ISSN)2701-3456
- Kosbab, Ben-Amon. "Trapped in Poverty: The Impact of Large-Scale Land Acquisitions on Education. A Geospatial Approach", MPPE No. 04/2024, Munich. (ISSN)2701-3456

Impressum:

ISSN: Editors: Associate Editors:

Managing Editor:

Contact:

2701-3456 Tim Büthe, Hanna Hottenrott Timm Betz, Sebastian Goerg, Michael Kurschilgen, Amy Pond, Sebastian Schwenen, Janina Steinert, Matthias Uhl Falk Bartscherer

Technicial University of Munich, Arcisstraße 21, 80333 München mppe@gov.tum.de, mppe@wi.tum.de https://www.wi.tum.de/mppe/ Twitter: @MunichPapers